Contents lists available at ScienceDirect

Applied Energy



A review of data-driven fault detection and diagnostics for building HVAC systems

Zhelun Chen^{a,*}, Zheng O'Neill^b, Jin Wen^a, Ojas Pradhan^a, Tao Yang^b, Xing Lu^b, Guanjing Lin^c, Shohei Miyata^d, Seungjae Lee^e, Chou Shen^e, Roberto Chiosa^f, Marco Savino Piscitelli^f, Alfonso Capozzoli^f, Franz Hengel^g, Alexander Kührer^g, Marco Pritoni^c, Wei Liu^h, John Claußⁱ, Yimin Chen^c, Terry Herr^j

^a Drexel University, Philadelphia, PA, United States of America

- ^b Texas A&M University, College Station, TX, United States of America
- ^c Lawrence Berkeley National Laboratory, Berkeley, CA, United States of America

^d The University of Tokyo, Tokyo, Japan

e University of Toronto, Toronto, Canada

^f Politecnico di Torino, Turin, Italy

g AEE Institute for Sustainable Technologies, Gleisdorf, Austria

h KTH Royal Institute of Technology, Stockholm, Sweden

ⁱ SINTEF Community, Trondheim, Norway

^j Intellimation LLC, Philadelphia, PA, United States of America

ARTICLE INFO

Keywords: Building HVAC Fault detection Fault diagnostics Fault prognostics Data imputation Feature selection Data-driven Machine learning Anomaly detection

ABSTRACT

With the wide adoption of building automation system, and the advancement of data, sensing, and machine learning techniques, data-driven fault detection and diagnostics (FDD) for building heating, ventilation, and air conditioning systems has gained increasing attention. In this paper, data-driven FDD is defined as those that are built or trained from data via machine learning or multivariate statistical analysis methods. Following this definition, this paper reviews and summarizes the literature on data-driven FDD from three aspects: process, systems studied (including the systems being investigated, the faults being identified, and the associated data sources), and evaluation metrics. A data-driven FDD process is further divided into the following steps: data collection, data cleansing, data preprocessing, baseline establishment, fault detection, fault diagnostics, and potential fault prognostics. Literature reported data-driven methods used in each step of an FDD process are firstly discussed. Applications of data-driven FDD in various HVAC systems/components and commonly used data source for FDD development are reviewed secondly, followed by a summary of typical metrics for evaluating FDD methods. Finally, this literature review concludes that despite the promising performance reported in the literature, data-driven FDD methods still face many challenges, such as real-building deployment, performance evaluation and benchmarking, scalability and transferability, interpretability, cyber security and data privacy, user experience, etc. Addressing these challenges is critical for a broad real-building adoption of data-driven FDD.

1. Introduction

1.1. Background

Building systems, including heating, ventilation and air conditioning (HVAC) systems, are usually subject to faults that can lead to undesirable performance, such as excessive energy waste, high maintenance costs, uncomfortable indoor thermal environments, and poor air quality. These faults refer to sensor failure, equipment failure, or faulty system operation. Studies have shown that 15%–30% of energy may be wasted due to building system faults and improper controls [1]. Therefore, fault detection and diagnostics (FDD) or automated fault detection and diagnostics (AFDD) as it is also commonly referred to, is crucial to ensure reliable system operation and avoid energy waste. To be specific, fault detection is defined as "determination that the operation of the building is incorrect or unacceptable in some respect" and fault diagnostics is defined as "identification or localization of the

https://doi.org/10.1016/j.apenergy.2023.121030

Received 29 December 2022; Received in revised form 18 March 2023; Accepted 21 March 2023 Available online 28 March 2023 0306-2619/© 2023 Elsevier Ltd. All rights reserved.





^{*} Corresponding author. *E-mail address*: zc98@drexel.edu (Z. Chen).

cause of faulty operation" [2]. It is reported that FDD users in the office and higher education market sectors of the United States were able to achieve 10% median energy savings annually with two-year simple payback period [3]. It demonstrates the high competitiveness of FDD systems as a profitable investment option in the building sector.

Over the past decades, many FDD methods have been developed. With the advancement of data science and the wide adoption of building automation systems (BAS) or other smart building technologies, data-driven FDD is gaining increased attention. Compared to the traditional expert knowledge/rule based approaches that are typically seen in commercial-off-the-shelf FDD products, data-driven FDD requires little or no *a priori* knowledge and has the potential to achieve high detection and diagnostic accuracy at relatively low cost [4].

Despite the rapid development of machine learning techniques, the market is slowly adopting data-driven FDD as an alternative or complement to the traditional rule-based approaches. A recent systematic survey of fourteen FDD tools in the United States showed that pure expert knowledge-driven methods for fault detection and diagnostics continue to dominate the market, though a few vendors are beginning to use data-driven methods [5]. In Australia, Wall and Guo [6] performed a survey of five FDD tools, two of which claimed to be equipped with machine learning techniques. Yet, details of the data-driven methods were not fully disclosed.

1.2. Data-driven FDD definition

Before a further discussion, it is necessary to provide a definition of the term "data-driven FDD" on the basis of existing literature. During the past two decades, more than twenty review articles concerning FDD in building energy system have been published, and many of them focused on classifying existing FDD approaches. However, the scope of what data-driven FDD encompasses varies. In the classical two-part review by Katipamula and Brambley [1,7], FDD studies were classified into quantitative model-based, qualitative model-based, or process history (data-driven) - based depending on how they approach the problem of fault diagnosis. This classification is loosely based on the classification employed by Venkatasubramanian et al. [8,9,10]. Generally speaking, quantitative and qualitative model-based approaches require a priori knowledge of the physical process, while process history-based approaches do not. Although this classification provided a good standing point at that time, Venkatasubramanian et al. [8] acknowledged that it can be confusing sometime as some methods were not one or the other. For example, some physical models require process data to tune the parameters (e.g., grey-box model) and some process history-based methods are quantitative (e.g., neural network) [11]. Despite its shortcomings, this classification scheme was adequate at the time when most methods were based on physical models. It is commonly adopted by other reviewers, such as Chen et al. [12]. With the rapid growth of data science in recent years, a number of recent FDD review articles have focused on data-driven FDD. Zhao et al. [13] conducted a review on artificial intelligence-based (AI-based) FDD methods for building energy systems. They proposed two classification schemes for fault detection and fault diagnosis, respectively. In both schemes, studies were classified into data-driven-based or knowledge-driven-based. For fault detection, data-driven-based methods included classification-based, unsupervised learning-based, and regression-based methods, while knowledgedriven-based methods included model-based and rule-based methods. For fault diagnosis, data-driven-based methods included classificationbased and unsupervised learning-based, while knowledge-driven-based methods included inference-based (e.g., Bayesian network, fuzzy logic) and diagnostic rule-based methods. Upon this classification, they further labeled a part of the methods as AI-based. For fault detection, the AI-based methods overlapped with the data-driven-based methods. For fault diagnosis, in addition to data-driven-based methods, inferencebased methods from the knowledge-driven-based side were also labeled as AI-based. Mirnaghi and Haghighat [14] conducted a review on FDD

of large-scale HVAC systems using data-driven methods. They classified FDD into model-based, data-driven, or knowledge-based. Model-based methods included first principle models and grey-box models. Data-driven methods were further divided into qualitative and quantitative methods. According to Mirnaghi and Haghighat [14], expert systems, fuzzy logic, pattern recognition, frequency analysis were data-driven qualitative-based methods. And statistical methods and neural networks were data-driven quantitative-based methods. Knowledge-based methods were defined as the combination of a qualitative part of the model-based method, including structural graphs, fault trees, or qualitative physics, and data-driven subcategory including fuzzy logic or expert systems.

From the literature, we have not seen any classification scheme that can perfectly classify the wide variety of FDD methods, especially if an FDD method includes multiple techniques. One of the main reasons is that there are no uniform definitions for terms such as data-driven, knowledge-driven, and model-based. In fact, (1) data often contains some levels of knowledge; (2) knowledge is typically presented as data; (3) the term model is often used to express a mathematical relationship between inputs and outputs (which may or may not be based on first principles). Another difficulty is that data could come from various sources, like real building systems, laboratory testing, or even simulation models. In addition, newly proposed methods are often combinations of techniques belonging to different categories. Therefore, in this paper, we will not focus on the categorization of various data-driven FDD methods, but rather on the process, systems studied, and evaluation metrics of data-driven FDD. We hope that this breakdown allows the readers to have a more holistic understanding of data-driven FDD to foster the development and market introduction of data-driven FDD products.

In most publications, the terms data-driven and machine learning are often used interchangeably and machine learning often is not strictly differentiated with multivariate statistical analysis. Therefore, we refer to those FDD methods that have at least one component, such as baseline modeling, fault detection, or fault diagnosis, which are built or trained from data via machine learning or multivariate statistical analysis methods, as data-driven FDD [15]. The effectiveness of the data-driven model is often limited by the training dataset. Therefore, the data quality, resolution, completeness, extensiveness (i.e., whether the data accounts for a variety of operating conditions), and uncertainty (i.e., measurements errors) are particularly important for data-driven FDD methods [1,13].

1.3. Literature gaps and organization of the paper

Despite extensive literature on FDD, a comprehensive review focusing on data-driven methods is still missing, especially one that meets the above definition and the scope. To further advance the development and market adoption of data-driven FDD, this paper provides a comprehensive review of the state-of-the-art data-driven FDD technologies from three main aspects: process, systems studied, and evaluation metrics. The goal of this review is to provide a comprehensive introduction for researchers and practitioners who are new to the topic. It also offers a systematic framework that can be used to categorize these methods, as well as research gaps and future work directions for experienced practitioners.

The content of the article is organized as follows. Section 2 presents the literature collection methodology and provides an overview of the collected papers. Section 3 describes the data-driven FDD process and summarizes the methods used in each step. In addition, it describes studies that have compared different FDD methods. Section 4 summarizes the systems that the data-driven FDD methods are applied to. These systems include, among others: air handling units (AHU), variable air volume (VAV) systems, chiller systems, variable refrigerant flow (VRF) systems. This section also lists the faults that have been identified in these systems and the different data sources. Section 5 discusses typical metrics for evaluating data-driven FDD methods. Section 6 discusses challenges and opportunities for future advancement of data-driven FDD. Section 7 concludes the review.



Fig. 1. Wordcloud of the reviewed papers' keywords.



Fig. 2. Breakdown of publications by publication year.

2. Literature identification and screening

The literature reviewed in this paper was collected mainly using a Google scholar based methodology known as Sub-keyword Synonym Subtopics Searching (SSSS: https://github.com/lz356/SSSS). This method aims to exhaust relevant papers by multiple searches with synonyms and subtopics sub-keywords. This method has been proven successful in other review papers, such as Zhang et al. [16,17]. Three sub-keyword groups were used to collect papers from 2005 to 2021. The first group contains: "fault detection", "fault diagnostics". The second group contains: "data driven", "data mining", "AI", "blackbox model", "anomaly detection", "data cleansing". And the third group contains: "building", "HVAC". For each search, SSSS took one word from each sub-keywords group to form a new keyword, and searched it in Google scholar. Besides the papers collected through SSSS, relevant papers from the authors' own collections and papers cited in the reviewed studies were also investigated. Fig. 1 shows a visual summary of the keywords contained in the reviewed literature.

Fig. 2 illustrates the number of published technical papers focusing on data-driven FDD from 2005 to 2021, with a detailed breakdown on the year of publication. The blue bars show the total number of papers published per year while the orange line displays the cumulative number of publications from 2005. The figure shows a gradual increase of the number of publications with a more noticeable rise in the last ten years. Fig. 3 illustrates the breakdown of the number of publications for the different journals. In particular, only the journals with at least 3 publications are included in this figure. The journal Energy and Buildings ranks the first with 32 papers. These figures, while they may not include every relevant article, generally reflect the current trends of data-driven FDD development.

3. Main steps of data-driven FDD process

Based on the reviewed literature, a data-driven FDD process can be broadly generalized as illustrated in Fig. 4. The first step is the data collection process, in which data are collected from the BAS, meter, or other data sources (e.g., IoT network). Following this, initial data cleansing such as missing data imputation and outlier detection are often performed. The data cleansing process can be carried out before or after the data are sorted into the data pool, depending on the framework of each individual FDD method. Besides data from a real building, simulated data from a digital twin or a simulation model can also be stored in a data pool for model training purposes. Before data is used for FDD development, additional preprocessing (e.g., feature selection, data reduction, data scaling) is often required to improve training performance. Section 3.3 reviews feature selection techniques in more details, as they are crucial for data-driven applications. For real-time implementation, a data-driven FDD method often compares incoming data (also referred to as snapshot data in the literature) with a baseline, which is further discussed in Section 3.4. Here a baseline refers to data or model that represent normal operation. In some cases, a baseline is not necessary. For example, a distributed method was applied to faulty sensor detection which referred to neighbor sensors instead of baselines [18,19]. The final steps of an FDD process are fault detection, fault diagnostics and fault prognostics. It should be noted that some FDD methods can perform fault detection and diagnostics simultaneously, whereas, some methods only perform fault detection without isolating the root causes [13]. Therefore, it is reasonable to discuss fault detection and fault diagnostics methods separately. Besides fault detection and diagnosis, one may also perform fault prognosis. Although this field is relatively underdeveloped, it is meaningful to include it in a general process discussion as it is closely related to detection and diagnosis. Methods used in each FDD step are discussed in the following subsections.

3.1. Data collection

Increasing advancements in building digitization, smart sensing and metering technologies have allowed a large amount of data to be collected and saved for monitoring, analyzing, and controlling of building systems. While modern buildings have considerably more sensing and actuation points than ever before, using them for better data-driven decisions is challenging both on a scientific and an engineering level [20]. Collecting data from a BAS is often the most time-consuming and labor-intensive process due to the fact that an individual building may have its unique BAS configurations, database structures, and datapoint naming conventions [21]. The use of standardized communication protocols, e.g., Building Automation and Control Network (BACnet), and metadata schemas, e.g., Project Haystack, Brick, and the recent ASHRAE 223P standard [22–24], can effectively ease the data collection process. Nevertheless, FDD applications that utilize metadata or semantic graph are still very few and underdeveloped [25].

3.2. Data cleansing

Often, datasets obtained from the BAS or other sources may be incomplete due to sensor and equipment failures, communication or transmission issues, data corruption, or inconsistencies due to sensor noise, thus, leading to loss of valuable information [26]. These issues exist in almost all kinds of sensors/automation systems and can significantly affect the FDD outcomes since most data analysis and statistical tools are not designed to handle incomplete data. Therefore, in this section, literature reported methods for missing data imputation are mainly discussed. Existing literature on data imputation methods for building data has been focused on univariate time series data using statistical methods [27–29] as well as nonlinear machine learning and deep learning [21,30–32]. More recently, ensemble methods that utilize

Energy and Buildings



Fig. 3. Number of papers for journal.



*including techniques for modeling metadata

Fig. 4. A general data-driven FDD process.

multiple imputation methods to make an improved prediction of the missing values have also been developed [33,34]. Their research on ensemble methods have shown good performance compared to single imputation methods for building system data. While most of the techniques reported for missing data imputation are focused on generic data-driven applications, there are few studies addressing the missing data issue for FDD purposes. For instance, Li et al. [35] proposed a filtering technique that uses both temporal and spatial information to reconstruct missing values, resulting in an improved FDD performance. Similarly, Wang et al. [36] utilized expectation–maximization (EM) algorithm to impute missing data prior to fault diagnosis. Their comparative study showed significant improvement when adding data imputation to their fault diagnosis model.

3.3. Data preprocessing

Data used for training FDD algorithms often need to be preprocessed beforehand to achieve desired training performance. This step may include feature selection, data reduction, data scaling, data transformation, data partitioning, etc. [37]. Typically, this step is performed offline (e.g., selecting some relevant features from historical data), and then the results of the preprocessing (e.g., the selected features), are applied to the snapshot data. This section mainly reviews the feature selection techniques.

Considering that there are hundreds or thousands of sensors from a BAS, often consisting of redundant measurements, the selection of informative and representative features strongly affects the performance of a data-driven FDD method [38]. Using all the features available within a massive dataset would cause models to be overfitted and increases the model complexity and computational cost [38,39]. Therefore, feature selection processes are often used to find the key inputs for a data-driven model used for the FDD process. Using only selected features instead of the entire dataset reduces the model complexity and model overfitting issues.

Different categories of feature selection methods such as filter, wrapper, embedded, and hybrid methods have been reported for FDD applications [40]. For instance, Li et al. [41] developed a novel filter method named information greedy feature filter (IGFF) to efficiently select and identify the most informative features from the AHU measurements based on mutual information. The selected method showed good performance when applied to the ASHRAE RP-1312 dataset [42, 43]. Filter methods are computationally fast and less prone to overfitting: however, it may fail to find the subset of features that results with the highest model accuracy [39]. On the other hand, wrapper method is an exhaustive searching method that trains and evaluates a specific model with different feature combinations, then selects the optimal combination with the best performance. Namburu et al. [44] implemented a simple genetic algorithm (SGA)-based wrapper to determine the optimal feature set for chiller FDD applications. Similarly, Yan et al. [45] and Mulumba et al. [46] incorporated auto-regressive models with exogenous variables (ARX) and Support Vector Machine (SVM) for chiller FDD application. Chen et al. [47,48] developed a partial least square regression and genetic algorithm (PLSR-GA) wrapper method to identify candidate features that represent the system performance under different operational modes. In their method, the GA searching process is used to facilitate the process of searching candidate features which are iteratively fitted to the PLSR model to evaluate the model performance. Although wrapper methods have shown good model performance, they are computationally expensive and susceptible to overfitting since multiple models are trained and evaluated [38]. Embedded methods that combine both the filter and wrapper methods are usually incorporated into a specific learning algorithm such as decision trees (DT) and random forest (RF). Yan et al. [49] applied a DT-based fault diagnostic for AHUs using a classification and regression tree (CART) algorithm where the features are automatically selected when training the model. Embedded methods that are not incorporated into the learning are categorized as hybrid methods. Han et al. [50, 51] used mutual-information-based filter and genetic-algorithm-based wrapper method to search for the important sensors in data-driven chiller FDD applications.

More recently, feature extraction methods directed at identifying and extracting interesting "localized" patterns within a timeseries to guide the feature selection process have also gained attention. For example, Zhang et al. [38] developed a novel framework that integrated feature extraction and selection for whole building FDD. Statistical feature extraction techniques are first employed in this framework to actively extract features from raw sensor data using various window sizes and statistical measures. Then, a hybrid feature selection algorithm that combines the filter and wrapper method is used to select the best feature combination. This framework achieves high model generalization since it considers diverse durations of fault behavior among different fault types. However, it should be noted that using a wrapper method may increase the computation time significantly if the number of features extracted is too high.

3.4. Baseline establishment

In order to enable FDD, a baseline which represents the normal system operation is often needed. Here, a baseline is defined as a status at which the building's operation is considered to be satisfactory, for instance, when the building has just gone through a commissioning process [47,48]. It is well-recognized that most existing buildings do not operate the way they are supposed to. For a real building, a true fault-free status is hardly achievable. Hence, a more realistic fault detection process is to detect when a building's status is significantly different from its baseline status [47,48]. Such baseline can either be generated by simulation [52] or constructed from historical (normal) data collected from a building.

Building HVAC systems present significantly different system dynamics under different operation modes (heating, cooling, etc.). For example, the components and their parameters engaged in a heating mode for an HVAC system is very different from those in a cooling mode. Both weather and internal loads could trigger the HVAC system to change its operation mode (e.g., from economizer control to cooling when outdoor air temperature increases). Hence, it is often challenging to differentiate system parameter/behavior variations triggered by weather and/or internal conditions from abnormalities triggered by faults in the system [47,48]. Since most data-driven FDD methods detect and diagnose faults by comparing real-time or quasi real-time operation data (snapshot data) with the baseline, it is important that the baseline is under the same operation mode as the snapshot data. The majority of the existing baseline studies focus on pattern recognition and motif discovery strategies to construct a baseline using outdoor weather conditions, daily internal load profiles and temporal association rules to ensure that the constructed baseline is under the same operation mode as the incoming snapshot data [47,48,53–58].

In addition to being a function of the HVAC system modes (e.g., heating, cooling), the baseline is also influenced by the specific control strategy (i.e., the detailed sequence of operation) implemented. A baseline model would ideally be able to incorporate and learn these changes instead of detecting them as faults.

3.5. Fault detection

Data-driven fault detection strategies that determine whether the HVAC system has failures or abnormal operations have shown great potential in efficiently characterizing system operations and developing accurate system models that are scalable, while also reducing engineering time and labor cost. A wide range of fault detection methods have been studied in the literature that can be categorized as supervised, semi-supervised or unsupervised methods [12–14].

Supervised methods for fault detection are trained using both normal operation and labeled fault data to identify whether the incoming data is faulty or fault-free. Supervised methods can be further categorized into classification methods or regression methods based on the type of model output. While classification methods such as SVM [44, 45,59-64] and DT [56,65-71] are used to predict whether the incoming data belongs to the fault or fault-free class, regression methods such as support vector regressions (SVR) [72,73] and neural networks (NN) [52,69,74-81] typically predict continuous variables, representing the system operation status, which is then compared to the baseline to identify any occurring faults. Both types of supervised methods have been widely used for fault detection in building HVAC systems. However, the challenge with supervised methods is in obtaining sufficient labeled fault data for training the models, often leading to imbalanced classes. Given the difficulty and the cost to label data, models used by supervised methods are often trained using data collected from older components or simulation models which can lead to lower detection accuracy and higher false alarm rates.

Alternatively, semi-supervised methods are more suitable when only limited labeled fault data is available [14]. Semi-supervised methods transfer unlabeled data into labeled classes by comparing the incoming data with normal operation, and updating the training set iteratively [82]. Although semi-supervised methods perform better when limited labeled fault data is available, semi-supervised methods have a higher computational cost than supervised learning [14].

Lastly, unsupervised methods that do not require fault labels have also been reported in the literature. These methods are helpful in discovering hidden correlations within a building's dataset and allow fault impact analysis to be made during the detection process. Some of the popular methods in this category are clustering algorithms [54, 65,68,75,83–91] and principal component analysis (PCA) [44,47,48,63, 92–102] which are used typically with pattern recognition and motif discovery methods. Since only fault-free data is required for deploying unsupervised methods, these methods are easier to develop and deploy for fault detection purposes.

3.6. Fault diagnosis

Identifying or localizing the root cause of a fault or anomaly is typically more challenging than detecting the anomaly, since different faults (e.g., malfunctioning hardware, software errors) can lead to the same symptom. Correctly diagnosing the root cause of a fault often requires a detailed knowledge of the HVAC configuration and control strategies, both of which are specific to a building. The literature has reported several inference and classification methods for fault diagnosis. Bayesian network (BN) models based on the conditional probability theorem that predicts the fault beliefs based on a set of observations are popular [103]. BN models can incorporate the system structure information through probabilistic conditional relations between faults and their symptoms. These probabilities can be updated after new observations (evidence) of the system are obtained. Further, by adding uncertainty factors for reasoning, the BN model can avoid incorrect diagnosis by avoiding under-responsiveness or overresponsiveness to evidence [104]. Successful implementation of BNs for both component-level and system-level fault diagnosis has been demonstrated in the existing literature. For instance, Zhao et al. [104, 105] proposed a component-level diagnostic BN for different faults in AHUs and chillers. These BNs were developed based on the fault patterns found in projects including NIST 6964, ASHRAE projects RP-1020 and RP-1312. Similarly, Xiao et al. [106] developed a BN based FDD method for diagnosing faults in VAV terminal units. At the systemlevel, Verbert et al. [107] developed a model-based BN method to diagnose system-level HVAC faults which included interdependencies between different components to carry out continuous fault diagnosis. More recently, Chen et al. [108] developed a discrete BN-based method for cross-level faults diagnosis in commercial buildings. Unlike continuous BNs which use continuous probability distributions in each node of the network, the continuous variables are discretized to represent fuzzy events in a discrete BN [109]. This makes modeling the BN parameters easier and more efficient, especially when obtaining parameters from expert knowledge and incomplete field data [110]. Alternatively, dynamic BN models that describe the temporal relationship of the fault states within each time slice have shown to be effective for fault diagnosis as well. Shi et al. [111] employed a dynamic BN using simulated data which successfully identified persistent as well as transient faults with reduced false positive rates. By carrying past information, a dynamic BN allows fault beliefs to accumulate over time, thus helps eliminate measurement errors and improve diagnosis accuracy [112].

Other diagnostic inference methods such as fault-trees that are based on a decision tree and multiple binaries of if-statements are usually time-consuming and may be highly dependent on domain expertise [14]. Alternatively, classification methods such as SVMs [46, 113-116] and ANNs [52,74,75] are also popular. These methods generally require a large amount of labeled data for model training, which can be challenging to obtain in real-world applications. To overcome the challenge of insufficient labeled data. Mivata et al. [52] proposed a NN-based method using convolutional neural network (CNN) trained on dynamic system simulation data of various fault types. Their method demonstrated good performance in learning fault behaviors from simulated data to identify both equipment-level and system-level faults in real data. Overall, NN-based methods have received increasing attention in recent years and show more accurate results than other classification methods [117]. However, the inference process behind the diagnosis of such black-box models often lacks transparency and interpretability [118].

3.7. Fault prognosis

Fault identification and diagnosis through FDD may not be sufficient in cases where critical functions of the systems may have already failed before the fault symptoms are observed, leading to excess operation and maintenance cost [119]. Additionally, some faults in building HVAC occur gradually and although do not produce a significant effect on the operation at the time, they may lead to considerable energy waste over time [14]. For example, if the zone temperature sensor drifts resulting in a higher than actual value, the controller will supply more chilled water to the cooling coil to reduce the zone temperature incurring in energy wastage. Other examples of gradual faults include coil fouling, duct leakage, decrease in fan efficiency, etc. It is estimated that over 20% of HVAC systems are running under early stage of gradual faults resulting over 15% in energy waste [120]. Therefore, data-driven fault prognosis approaches, which refers to identifying impending faults ahead of time and estimating how soon a fault may occur by analyzing historical or real-time measurements for predictive maintenance and repair schedules, is essential for ensuring the safety, stability and for increasing the lifespan of HVAC systems. As an example for predictive maintenance, the maintenance program could be based on the performance degradation of a heat pump system. Typical key performance indicators for such an approach are the coefficient of performance (COP) of the heat pump or the system efficiency index (SEI), which is the ratio of the measured COP and the maximum COP at the same specified reference temperatures [121,122].

Data-driven fault prognosis methods have been gaining attention from different industrial sectors in recent years [119], however, development of fault prognosis strategies for HVAC systems is still in its infancy. Of the existing literature, Yang et al. [123] employed a text mining approach based on operator logbooks to produce multiple high-level metrics such as failure probabilities and mean-time to failure. Similarly, Yang et al. [124] developed regression tree models for estimating time-to-failure for chillers and boilers in a central heating and cooling plant. Ahmad and Atta [125] studied motor failures using electric current predictions and Wang et al. [126] developed an algorithm based on particle filters to estimate the remaining useful life (RUL) of heat exchangers. Their proposed method demonstrates its effectiveness in predicting both natural and transient degradations. Using their method, about half of the modeled failure events were accurately predicted in the distributed control system of the plant. Using typically available BAS data, Yan et al. [127] developed a Hidden Semi-Markov Model (HSMM)-based method to efficiently estimate the RUL of an AHU and its component. Their approach was evaluated using ASHRAE RP-1312 data for both single and multiple fault cases.

The strength of data-driven models for fault prognosis has been demonstrated in many industrial sectors. For example, Recurrent Neural Networks (RNNs) that can exploit temporal correlations in the data by using a feedback loop enables accurate predictions of time series. Such data-driven models for fault prognosis have been investigated in process operations [128] and predictive maintenance of machines [129, 130]. Besides RNN, Autoencoder (AE) [131] and Restricted Boltzmann Machine (RBM) [132] approaches are also applied in recent data-driven models for fault prognosis. Those architectures show great potential for fault prognosis of building HVAC systems.

3.8. Summary of data-driven FDD methods

Table 1 presents ten data-driven methods used for fault detection and diagnosis that are commonly seen in the reviewed papers. These methods are: Clustering, Decision Tree (DT), Principal Component Analysis (PCA), Support Vector Machine (SVM), Support Vector Regression (SVR), Neural Network (NN), Bayesian Network (BN), Hidden Markov Model (HMM), Generative Adversarial Network (GAN), and Ensemble Learning. Notice that the papers listed for each method serve as examples and are not all-inclusive.

Among all the reviewed papers, some compared different datadriven FDD algorithms in their studies. These comparison studies are illustrated in Table 2. For each comparison study, Table 2 presents information about (1) whether the study focuses on fault detection or fault diagnosis; (2) whether the studied method is supervised or unsupervised; (3) the HVAC system/component that the FDD method is applied to; and (4) the specific algorithms being compared. It can be seen that most compared algorithms belong to the same category. For example, Amruthnath and Gupta [133] applied and compared unsupervised methods for early fault detection in exhaust fans. Asgari et al. [80] compared two supervised models, CNN and RNN, to predict multiple simultaneous failures in the data center cooling units. There are limited studies that compare the performance between different categories (e.g., expert rule-based vs data-driven, supervised vs unsupervised).

4. Systems studied with data-driven FDD

Data-driven FDD methods have been reported to be applied to many HVAC components and subsystems for various types of faults. This section summarizes (1) the systems that data-driven FDD have been applied to; (2) the identified faults associated with the systems; and (3) the main data source utilized when developing and evaluating a datadriven FDD method. Table 3 presents the detailed information that is being discussed in this section.

4.1. Faulty systems and identified faults

Several studies discussed HVAC fault categories [120,183,184]. Based on these discussions, HVAC faults can be categorized as hardware and software faults by the types of component. Hardware faults further include equipment faults, sensor faults, and controlled device (including actuator) faults. Software faults further include controller faults (e.g., unstable control), human faults (operator faults) and control logic errors. Fig. 5 illustrates the workflow of these categories. Reviews Table 1

Commonly used data-driven methods for fault detection and diagnostics.			
Method	Reference		
Clustering	Capozzoli et al. [65], Du et al. [75], Narayanaswamy et al. [134], Fan et al. [83], Miller et al. [54], Cheng et al. [84], Li		
	et al. [35,85], Dey et al. [86,87], Aguilar et al. [88], Gunay and Shi [89], Xu et al. [90], Zhou et al. [91], Liu et al. [68]		
DT	Capozzoli et al. [65], Yan et al. [49], Li et al. [135], Capozzoli et al. [66], Piscitelli et al. [56], Tesfay et al. [67], Liu		
	et al. [68], Piscitelli et al. [69], Chiosa et al. [70,71]		
PCA	Wang and Qin [92], Namburu et al. [44], Du and Jin [93,94], Hu et al. [95], Li and Wen [96,97], Cotrufo and		
	Zmeureanu [98], Hu et al. [99], Yan et al. [136], Shi et al. [100], Li and Hu [101], Montazeri and Kargar [63], Zhou		
	et al. [102], Chen et al. [47,48]		
SVM	Namburu et al. [44], Liang and Du [59], Li et al. [113], Han et al. [51], Dehestani et al. [60], Yan et al. [45], Beghi		
	et al. [61], Beghi et al. [114], Mulumba et al. [46], Fan et al. [115], Madhikermi et al. [62], Montazeri and Kargar		
	[63], Li et al. [64], Lee et al. [116]		
SVR	Zhao et al. [72], Liu et al. [73]		
NN	Cho et al. [137], Hou et al. [138], Du et al. [139], Magoulès et al. [74], Dehestani et al. [60], Du et al. [75], Jones		
	[76], Guo et al. [140], Shi et al. [100], Madhikermi et al. [62], Lee et al. [77], Shahnazari et al. [141], Sipple [78], Xu		
	and Chen [79], Miyata et al. [52], Piscitelli et al. [69], Asgari et al. [80], Han et al. [142], Liu et al. [143], Taheri		
	et al. [144], Liao et al. [81]		
BN	Wall et al. [145], Zhao et al. [146], Dong et al. [147], Xiao et al. [106], Zhao et al. [104,105], Verbert et al. [107],		
	Taal et al. [148], Wang et al. [149], Li et al. [118], Chen et al. [108]		
HMM	West et al. [150], Yan et al. [127,151]		
GAN	Zhong et al. [152], Yan et al. [82,153], Li et al. [154]		

Araya et al. [155], Fan et al. [156], Zhong et al. [152], Han et al. [157]

Table 2

Ensemble learning

A collection of the comparison studies using different data-driven FDD algorithms.

References	Detection	Diagnosis	Proposed FDD type	System	Algorithm comparison
Namburu et al. [44]		x	Supervised	Chiller	SVM, PCA, PLS
Capozzoli et al. [65]	x		Unsupervised	Whole building	CART, K-Means, DBSCAN
Yan et al. [45]		x	Supervised	Chiller	ARX+SVM, LinReg+SVM, ARX+NN, SVM
Jones [76]	x	x	Supervised	AHU	Detection: ART, LAPART, NN, SVM,
					Rule-based; Diagnosis: SVM, LAPART
Mulumba et al. [46]		х	Supervised	AHU	NB, BN, RBF, MLP, SVM, RF
Li et al. [158]		х	Supervised	Chiller	SVM, DT, NN, AB, QDA, LogReg
Guo et al. [159]		х	Supervised	VRF	DBNs w/ various settings
Amruthnath and Gupta [133]	x		Unsupervised	Exhaust fan	PCA, T2 statistic, Hierarchical clustering,
					K-Means, Fuzzy C-Means
Li et al. [135]		х	Supervised	VRF	CART, RF, GBM
Li and Hu [101]	х	х	Unsupervised	Chiller	PCA, DBSCAN
Dey et al. [87]		х	Supervised	FCU	SVM, NN
Zhou et al. [160]		х	Supervised	VRF	DT, SVM, clustering, SNN, DNN
Ebrahimifakhar et al. [161]		x	Supervised	RTU	LogReg, LDA, QDA, KNN, Bagging, RF,
					AB, XGB, SVM
Yan et al. [82]	х	х	Semi-supervised	Chiller	SVM, RF, DT, BN, KNN, LogReg w/ and
					w/o GAN
Yan et al. [153]	х	х	Semi-supervised	AHU	RF, SVM, MLP, KNN, DT w/ and w/o
					GAN
Shohet et al. [162]	х		Supervised	Boiler	KNN, DT, RF, SVM
Han et al. [157]		х	Supervised	Chiller	Ensemble learning, KNN, SVM, RF
Li et al. [163]		х	Semi-supervised	Chiller	Semi-GAN, NN, DBN, LS-SVM
Asgari et al. [80]	x	х	Supervised	AC unit	CNN, RNN
Wang et al. [164]		х	Supervised	Chiller	SVM, BN, SVM+BN
Taheri et al. [144]		x	Supervised	AHU	deep RNN, RF, GB

on the impact of faults in each category are out of the scope of this paper. For interested readers we recommend the following studies on impact of faults: equipment faults from Mirnaghi and Haghighat [14] and Rogers et al. [11], sensor faults from Zhang et al. [185], controlled device faults from Weimer et al. [186], human faults from Torabi et al. [187], and fault impact evaluation framework from Lu et al. [188]. Most published papers focus on single fault, despite the fact that simultaneous faults could occur in real systems. Some recent studies such as Hu et al. [189], Hu and Yuill [190] start to discuss impacts from simultaneous faults.

Fault detection and diagnostics can be implemented at the building level, system/subsystem level (e.g., an air handle unit), and/or component level (e.g., a damper) [183]. Building-level FDD aims to detect and diagnose the occurrence of non-optimal operational patterns by identifying anomalous energy trends in the building energy consumption time series [54,66]. At this scale, classification, regression and pattern recognition techniques are employed to estimate the baseline for the detection of anomalies while sub-meter load data are used to infer the root cause of anomalies at whole-building level [70,71].



Fig. 5. Fault distribution in an HVAC system. *Source:* Expanded from Yu et al. [120].

System	Data source	Reference
AHU or AHU-VAV	lab data	Cho et al. [137], Hou et al. [138], West et al. [150], Wall et al. [145], Dehestani
		et al. [60], Li and Wen [96,97], Mulumba et al. [46], Zhao et al. [105], Yan et al.
		[49], Zhao et al. [104], Yan et al. [127], Zhong et al. [152], Li et al. [35], Piscitell
		et al. [56], Yan et al. [153], Dowling and Zhang [165], Li et al. [118], Taheri et al.
		[144], Liao et al. [81]
	real data	Wang and Qin [92], West et al. [150], Dong et al. [147], Narayanaswamy et al.
		[134], Cheng et al. [84], Madhikermi et al. [62], Gunay and Shi [89]
	sim data	Wang and Qin [92], Liang and Du [59], Du and Jin [93,94], Li et al. [113], Du
		et al. [75,139], Xiao et al. [106], Jones [76], Yan et al. [127,136], Verbert et al.
		[107], Turner et al. [166], Shahnazari et al. [141], Lee et al. [77], Yan et al. [151]
		Montazeri and Kargar [63]
Chiller	lab data	Choi et al. [167], Namburu et al. [44,168], Han et al. [50,51], Zhao et al. [146],
		Yan et al. [45], Zhao et al. [169], Beghi et al. [61,114], Li et al. [85,158], Beghi
		et al. [170], Fan et al. [115], van de Sand et al. [171], Wang et al. [149], Han et al
		[157], Liu et al. [172], Yan et al. [82], Li et al. [64], Yan [173], Li et al. [163], Har
		et al. [142], Li et al. [154,174], Wang et al. [164], Liu et al. [143], Gao et al. [175]
	real data	Hu et al. [95], Zucker et al. [176], Dong et al. [147], Cotrufo and Zmeureanu [98],
		Cheng et al. [84], Hu et al. [99], Li et al. [177], Li and Hu [101], Lee et al. [116]
	sim data	Du and Jin [93], Zhao et al. [72], Miyata et al. [52]
Boiler	sim data	Verbert et al. [107], Shohet et al. [162]
AC unit	sim data	Asgari et al. [80]
RTU or RTU-VAV	lab data	Li et al. [154]
	sim data	Ebrahimifakhar et al. [161]
FCU	real data	Dey et al. [86], Dey et al. [87]
VRF	lab data	Guo et al. [140], Shi et al. [100], Li et al. [135], Guo et al. [159], Liu et al.
		[73,178], Zhou et al. [160], Zeng et al. [179], Zhou et al. [102]
Heat pump	lab data	Zogg et al. [180], Tesfay et al. [67]
	real data	Taal et al. [148]
District heating	real data	Xue et al. [181]
Whole building	real data	Jacob et al. [182], Capozzoli et al. [65], Fan et al. [83], Miller et al. [54], Araya
		et al. [155], Capozzoli et al. [66], Fan et al. [156], Gunay and Shi [89], Sipple [78]
		Xu and Chen [79], Aguilar et al. [88], Zhou et al. [91], Xu et al. [90], Piscitelli
		et al. [69], Chiosa et al. [70], Liu et al. [68], Chen et al. [47], Chiosa et al. [71],
		Chen et al. [48,108]
	sim data	Magoulès et al. [74]

 Table 3

 Summary of systems studied with data-driven FDD and associated data sources

Not all potential faults are studied in the existing research papers and reports, and different HVAC systems are focused for different building sizes. For large size buildings, FDD is often applied to AHU-VAV systems, fan coil units (FCU), chillers, and boilers. Yet for small and medium sized buildings, FDD is usually applied to heat pumps and window air conditioners. From the reviewed papers, the most popular research subjects are secondary AHU-VAV systems (35%) and chillers (32%), followed by whole-building studies (17%) and VRFs (7%). The top two faulty systems (i.e., secondary systems and vapor compression systems) and the corresponding classification of the faults in these systems (summarized from the references listed in Table 3) are illustrated in Fig. 6. Secondary systems can provide required heating and cooling for multiple zones. Its actuator faults and equipment faults are typically studied, including OA/RA/SA/VAV dampers, cooling/heating coil valves, SA/RA fans, and air ducts. Vapor Compression Cycle (VCC) systems, consisting of a set of components (compressors, heat exchangers, and expansion devices), are included in a wide range of equipment, from small split air conditioners that are usually used in residential houses, to large central systems such as chillers that provide significant cooling/heating capacity to serve an entire office building, hospital, or campus buildings. Among them, chiller faults are the most widely studied with regards to data-driven methods. Those faults can be classified as local faults (e.g., condenser fouling, reduced condenser water flow, non-condensable in the refrigerant, reduced evaporator water flow, etc.) and system faults (e.g., refrigerant leak/undercharge, refrigerant overcharge, excess oil, etc.) [149]. Furthermore, a proper time resolution of the measurement data should be chosen depending on which system level is of interest. For example, determining the COP of a heat pump with on/off control might only need data with 15-minute resolution, whereas higher resolution (e.g., 1-minute) can reveal operational issues, such as too frequent heat pump cycling or an unreasonable defrost cycle schedule. These control logic issues can

lead to a poor energy performance even though all components work properly.

Although most of the literature on FDD focuses on the system/ component level, FDD at whole-building level has increasingly attracted the interest of researchers. The influence of several factors such as building dynamics, external climatic conditions, system operating schedules and occupant comfort requirements can determine the occurrence of numerous building energy consumption patterns that are not always easy to be recognized. In addition, in most of the reallife applications, only few and aggregate variables pertaining to the whole-building and main sub-loads energy consumption are available, which increases the complexity in conceptualizing FDD processes at the building level. In this context, the main objectives of a buildingscale FDD process are (1) the recognition of typical patterns in the whole-building energy consumption time series, (2) the detection of infrequent/anomalous patterns, and (3) the diagnosis of the detected anomalies by inferring the presence of anomalous patterns at the subload level. All three objectives require expert domain knowledge on how the building and its energy systems work and are operated. Different from applications at the system level, anomalies at a higher level of analysis are difficult to be classified and generalized due to the lack of ground-truth datasets for conducting both detection and diagnosis. As a consequence, for applications at whole building level, unsupervised learning through pattern recognition techniques is the most employed approach [47,48,108].

4.2. Data source

For data-driven FDD method development, data, such as labeled normal and fault data, are needed for training and method evaluation purposes. These data can be supplied from simulation, laboratory experiments, and field measurements from a real building. Among the papers reviewed, excluding those in which the data source was not



Fig. 6. Fault classification of HVAC systems.

explicitly stated, 48% use lab experiment data, 20% use simulation data, 32% use real building data. The coupling between the data source and the system type is further analyzed in Fig. 7 which shows that the majority of whole building applications rely on real field measurement data while system-level VRF, AHU and Chiller applications mainly rely on laboratory data.

Among all the reviewed studies, lab experiment data were most commonly adopted. They had been applied in tree-structured learning FDD of chillers [158], GAN-based FDD of AHUs [153], deep learningbased FDD of VRF systems [159] and so on. However, there are limited examples of publicly available datasets that have verified groundtruth information on the presence and absence of faults. Based on the reviewed studies, four research projects offer valuable public available experimental data. The first two are ASHRAE Project 1043-RP data and ASHRAE Project 1312-RP data, which have been widely used in chiller and AHU FDD studies, respectively. The third is the building fault detection data to aid diagnostic algorithm creation and performance testing. The fourth is the LBNL fault detection and diagnostics datasets [191]. Notice that in many cases, a laboratory testbed is a real building with real HVAC system, but it is not occupied and is solely used for testing purposes, such as the ASHRAE 1312 testbed [192]. Details about these four datasets are further discussed below.

ASHRAE Project 1043-RP data [193]: The experiments were conducted on a R134a refrigerant centrifugal chiller with 90 tons (316 kW) cooling capacity. Nine typical chiller faults were artificially implemented to the test chiller, including reduced evaporator water flow fault, reduced condenser water flow fault, a combination of these two faults, condenser fouling fault, refrigerant leakage fault, refrigerant overcharge fault, excess oil in the compressor fault, non-condensables in refrigerant fault, and a defective pilot valve fault. Each fault was implemented



Fig. 7. Alluvial plot for the analyzed literature classified by data source and system type.

at four severity levels ranging from 10% to 40%. At each level of each fault, the experiment was conducted under 27 different operating conditions for about 14 hours. The operational data of 64 variables (e.g., temperature, pressure, rate of flow, power) were collected at 10 seconds and 2 minutes intervals. In each operating condition, the experimental test was first carried out to reach the steady state for about 30 minutes. Next, the steady state test was continuously performed for 15–25 minutes for each operating condition.

- ASHRAE Project 1312-RP data [42,43,192]: A series of experiments were performed on two identical systems (system A and B). Each system includes a multizone AHU-VAV (AHU-A and AHU-B) that serves four building zones (three external and one internal). The AHU-VAV and building zones from each system are exactly identical to the other system. AHU-A was always operated under normal conditions while AHU-B was used to implement different faults. Seventeen faults were tested, including equipment faults, control device faults, sensor faults, and controller faults. Each fault was tested at multiple fault intensity levels in three seasons spring, summer, and winter. At each level of each fault, the experiment lasted for one day and the operational data of 160 variables were collected at 1-min intervals.
- Building fault detection data to aid diagnostic algorithm creation and performance testing [194]: Two experimental test datasets were created for a single-zone AHU-CAV and a single-zone AHU-VAV serving a real building zone. The experiments include the following AHU faults at multiple fault intensity levels: outside air damper stuck fault, heating/cooling coil valve stuck/leakage fault, and outdoor air temperature sensor bias fault. Another experimental test dataset was generated for RTU-VAV system faults: condenser fouling, HVAC or lighting system setback error (delayed onset or early determination), excessive infiltration, no overnight HVAC or lighting setback, and thermostat temperature bias. In all three datasets, 24 hour operational data of normal and fault test cases were collected at 1-minute intervals.
- LBNL fault detection and diagnostics datasets [191]: The project includes an RTU experimental test dataset, covering faults of incorrect economizer setpoint, outside air damper stuck, and supply air temperature sensor bias. The experiments were performed in spring, summer, fall, and winter seasons and all data use 1-minute measurement frequency.

Simulated data generated from non-proprietary physical modelbased simulation software were also used in the reviewed studies. A detailed review of current state-of-the-art for the fault modeling of HVAC systems in buildings, including fault model, fault occurrence probability, and fault simulation platform, was provided in Li and O'Neill [195]. For instance, Du et al. [75] validated a combined neural networks algorithm to detect AHU abnormities using a TRNSYS-based simulation testbed. Lee et al. [77] employed EnergyPlus software to generate different types of fault operation behavior data to serve as references for a deep-learning FDD algorithm for AHUs. Montazeri and Kargar [63] used simulated data from HVACSIM+ software to train and test SVM and PCA FDD methods. Shahnazari et al. [141] modeled an AHU-VAV system using Modelica to demonstrate the ability of recurrent neural networks-based FDD algorithm. In Lu et al. [188], a comprehensive fault impact analysis and robustness assessment of the high-performance control sequences from ASHRAE Guideline 36 was conducted using a Modelica-based fault modeling framework, where 359 fault scenarios, including faults of sensor, duct and pipe, valve and damper, HVAC equipment (e.g., coil fouling, fan mode degradation, etc.), control, design and construction, were simulated. Since these simulation software are originally designed to simulate fault-free operations, fault generators or similar means are needed to simulate faulty operations. Simulating faults using these simulation software sometimes results in unexpected numerical difficulties such as long simulation time, inaccurate results, or even crashing of the simulation program. When modeling faults, it is important to avoid some common causes of numerical difficulties, such as numbers that are beyond the

computer precision [196] and discontinuous functions [197]. When it comes to the generation of simulation data for large-scale HVAC systems, an advanced numerical solver that is both efficient and robust shall be considered. For example, when simulating the HVACSIM+ model developed for the ASHRAE Project 1312-RP [192], an advanced nonlinear equations solver [198] can reduce the simulation time by about 70% while maintaining the same level of robustness as the default solver, and even more for larger scale systems. In terms of opensourced simulation datasets, LBNL FDD datasets [191,199,200] include large simulated datasets with verified information on the presence and severity of faults spanning seven HVAC systems and configurations: a single-duct AHU, a RTU, a dual-duct AHU, a series fan-powered VAV unit, a parallel fan-powered VAV unit, a fan coil unit, a chiller plant, and a boiler plant. HVACSIM+ and Modelica-EnergyPlus co-simulation were employed to carry out simulations of more than 250 faulted or fault-free condition states (e.g., mechanical faults, control sequence faults, sensor faults, etc.) over a full year of operation. Each dataset includes from 20 to more than 100 data points which are described with the Brick schema [22].

Although 32% of the reviewed studies use real building data, half of the studies focus on the detection of energy anomalies that only require energy consumption data. For example, Zhou et al. [91] tested a hierarchical clustering method to identify anomalies in daily energy consumption with the chiller plant power consumption data. Publicly available power consumption datasets that can be used to validate anomaly detection algorithms are rare. The competition "Power Laws: Detecting Anomalies in Usage" offers a few datasets with handlabeled anomalies corresponding to different types of building sites from different geographies [201]. Field measurements with labeled faulty operation data are more challenging to obtain since manual investigation and/or maintenance record are typically needed to confirm the occurrence of a naturally occurred fault. LBNL FDD datasets [191] also include months of field measured RTU data for a compressor control fault and a refrigerant undercharging fault. Measurements from the commissioning phase of a real building often contains faults and can thus be a valuable part of a training dataset. Documentation of changes in the system throughout the commissioning process is of utmost importance to understand what was a normal or faulty operation. Among studies that utilized real building operation data, Gunay and Shi [89] demonstrated a cluster analysis-based anomaly detection method with a year's worth of BAS data from 247 thermal zones and an air handling unit. Lee et al. [116] implemented AI-FDD on 14 chillers for a 1 year test to obtain the real-world false alarm rejection rate. Chen et al. [47,48] developed a weather and schedule informationbased pattern matching (WPM) and feature-based principal component analysis (FPCA) method to detect whole building level faults using 30 day BAS data from a multi-used campus building.

In addition to the labeled time-series system operation or energy use data under normal and fault conditions, other data sources have also been utilized to support the data-driven FDD algorithms. For example, expert knowledge, maintenance records, building information models (BIM), and real-time occupancy data. Expert knowledge has been integrated into data-driven FDD approaches to (1) detect outliers in the data preprocessing step [83], (2) develop, select, and interpret the characteristic features of faults [107,170,181,202], and (3) support the selection of layers, nodes, and parameters in the BN-based or treestructured FDD algorithms [108,146,158]. Maintenance records were utilized to label the ground truth of field measurements, i.e., whether the collected data contain faults or not [116,148]. The BIM was integrated into model-based FDD to provide building design information (e.g., architecture geometry and building equipment information) [147, 203,204]. Real-time building occupancy data from internet of things sensors were employed as an additional data stream to detect the degraded building operation performance [84].

Table 4

Summary of evaluation metrics for data-driven FDD.				
Evaluation metrics	References			
General evaluation metrics (e.g., Eqs. (1)–(8))	Zogg et al. [180], Jacob et al. [182], Shohet et al. [162], Hu et al. [95], Zhao et al. [72], Sun et al. [205], Zhao et al. [146], Yan et al. [45], Du et al. [75], Narayanaswamy et al. [134], Jones [76], Beghi et al. [114], Li et al. [177], Araya et al. [155], Turner et al. [166], Verbert et al. [107], Guo et al. [140], Liu et al. [73], Guo et al. [159], Shi et al. [100], Zhong et al. [152], Liu et al. [172], Zhou et al. [160], Li et al. [163], Guo and Rasmussen [206], Zhou et al. [102], Piscitelli et al. [69],			
Classification problem metrics	Chen et al. [47,48,108] Choi et al. [167], Namburu et al. [168], Du and Jin [93], Namburu et al. [44], Liang and Du [59], Li et al. [113], Dehestani et al. [60], Magoulès et al. [74], Mulumba et al. [46], Yan et al. [49], Li et al. [158], Yan et al. [127], Li et al. [135], Lee et al. [77], Liu et al. [178], Fan et al. [49], Li et al. [35], Montazeri and Kargar [63], Ebrahimifakhar et al. [161], Yan et al. [153], Zeng et al. [179], Aguilar et al. [88], Yan et al. [82], Dowling and Zhang [165], Dey et al. [87], Gao et al. [175], Yan [173], Lee et al. [116], Chiosa et al. [70], Li et al. [154,174], Wang et al. [164], Liu et al. [143]; Beghi et al. [61], Sipple [78], Asgari et al. [80]			
Statistical significance tests	Han et al. [157], Dey et al. [86]			

5. Evaluation metrics for data-driven FDD

It is critical to evaluate and quantify the performance and effectiveness of data-driven FDD methods by using dedicated metrics. In this section, evaluation metrics adopted from the reviewed literature for data-driven FDD are summarized and discussed. Corresponding studies for each evaluation metric are illustrated in Table 4. The collected metrics can be broadly classified into three categories: general evaluation metrics for FDD applications, evaluation metrics for data-driven classification problems, and statistical significance tests that assist the evaluation of classification problems.

General evaluation metrics

Lin et al. [207] summarized the evaluation metrics for general FDD applications. To assess the performance of a fault detection problem, the evaluation metrics include the true positive rate (TPR), true negative rate (TNR), false positive rate (FPR), false negative rate (FNR), and no detection rate (NDR), which are shown in Eqs. (1)–(5). To assess a fault diagnosis method, the evaluation metrics often include the correct diagnosis rate (CDR), the misdiagnosis rate (MDR), and the no diagnosis rate (NDgR), as listed in Eqs. (6)-(8). Note that each evaluation metric mentioned above focuses on a specific aspect of the FDD problem, and thus relying solely on one metric can be misleading. For example, in a dataset with a large number of fault-free samples and a small number of fault samples, TNR can be misleading, since an algorithm that simply predicts fault-free for all samples can still achieve a high TNR. Therefore, it is important to evaluate these metrics together. To provide a clear understanding of these metrics, Fig. 8 visualizes the relationship of the terms used in these equations in a confusion matrix, which will be further discussed in the section below.

$$TPR = \frac{TP}{TP + FN}$$
(1)

$$TNR = \frac{TN}{TN + FP}$$
(2)

$$FPR = \frac{FP}{TN + FP}$$
(3)

$$FNR = \frac{FN}{TP + FN}$$
(4)

$$NDR = \frac{ND}{TP + TN + FP + FN}$$
(5)

$$CDR = \frac{CD}{TP + FN}$$
(6)

$$MDR = \frac{MD}{TP + FN}$$
(7)

$$NDgR = \frac{NDg}{TP + FN}$$
(8)



Fig. 8. Example of a confusion matrix. Given the predicted fault types on the columns and the actual fault types on the rows, the matrix shows the correct diagnosis (CD), misdiagnosis (MD), true positives (TP), true negatives (TN), false positives (FP) and false negatives (FN).

Classification problem metrics

An FDD problem is essentially a classification problem; fault detection is a binary classification problem, while fault diagnosis is a multi-class classification problem. The general metrics described above are often combined visually or quantitatively into a classification problem metric for a more in-depth evaluation of data-driven classifiers. These classification problem metrics include confusion matrix, accuracy of correct predictions, F-measure (or F-score), Receiver Operator Characteristic (ROC), and Area Under the Curve (AUC).

Confusion Matrix. A confusion matrix is a visualization of prediction results for a classification model [208]. It depicts the degree of algorithm confusion within different classes and is independent of a concrete classification algorithm [157]. Each matrix element represents the test observations, with the actual (true) class in rows and the predicted class in columns. The diagonal elements show the correct predictions while the off-diagonal elements show the incorrect predictions and how they were misclassified. For a 2 by 2 confusion matrix, two rows and two columns report the number of true positive, false negative, false positive, and true negative. Ebrahimifakhar et al. [161] visualized the classification performance of different algorithms in diagnosing faults for packaged rooftop units.

Accuracy of Correct Predictions. The overall accuracy of correct predictions is defined as the number of correct predictions (i.e., the diagonal elements of the confusion matrix) divided by the total number of observations, as shown in Eq. (9). Montazeri and Kargar [63] compared the diagnostic results of the proposed FDD method (i.e., KPCA and RBF neural network) and previous studies by visualization of the confusion matrix and using the accuracy. Overall, it is a simple and intuitive measure, yet it may fail on classification problems with a skewed class distribution.

$$Accuracy = \frac{TP + TN}{TP + TN + FP + FN}$$
(9)

F-Measure. F-measure [209] is a comprehensive performance metric to evaluate the quality of a classifier which considers the class-specific performance, as shown in Eq. (10). F-measure ranges from 0 to 1. The larger the F-measure is, the better the comprehensive performance of the classification model is [210]. In the equation, "Precision" refers to the proportion of correctly diagnosed samples in all positive samples, while "Recall" refers to the proportion of correctly diagnosed samples in the true samples.

$$F = 2 \cdot \frac{\text{Precision} \cdot \text{Recall}}{\text{Precision} + \text{Recall}}$$
(10)

 $Precision = \frac{TP}{TP + FP}$ (11)

$$\text{Recall} = \frac{\text{TP}}{\text{TP} + \text{FN}}$$
(12)

ROC and AUC. The ROC curve is a graph showing the performance of a classification model at all classification thresholds [211] which plots two parameters: TPR and FPR. One potential drawback of the ROC curve is that it can be difficult to interpret if there are many decision thresholds. This is because each point on the curve represents a different tradeoff between the TPR and FPR, and it may not be immediately clear which point represents the best overall performance of the algorithm. AUC measures the entire two-dimensional area underneath the entire ROC curve from (0,0) to (1,1). A higher AUC [212] indicates that the model performs better in distinguishing between positive and negative classes. Sipple [78] used AUC to compare the anomaly detection performance of various models on predicting failures of HVAC equipment.

Statistical significance tests

Statistical significance (or hypothesis) tests can aid in comparing the performance of different classification models. The purpose of statistical significance testing is to help gather evidence of the extent to which the results returned by the aforementioned evaluation metrics are representative of the general behavior of the classifiers. However, it is noted that significance testing never constitutes a proof that the observation is valid. It provides added support for the observations. Ultimately, demonstrating that a new FDD algorithm performs better than a reference algorithm requires a combination of statistical tests, evaluation metrics, and validation experiments. The frequently used significance tests, for data-driven FDD, include the t-test [213], McNemar's Test [214], Wilcoxon's signed-Rank Test [215], Friedman Test [216], Nemenyi Test [217], etc. Han et al. [157] used Friedman Test and Nemenyi Test to evaluate the performance of different classification models on diagnosing the chiller faults.

6. Future challenges and opportunities

Although research on data-driven FDD has made great advancements in recent years as discussed above, its broad market adoption remains limited [5]. In this section, we discuss some of the ongoing efforts and challenges to further the development and market adoption of data-driven FDD.

Real-building deployment

As discussed in Section 4.2, although much research has been conducted to develop and implement data-driven FDD methods for building systems, most of the studies developed or validated their datadriven methods in simulated environments, in laboratory settings, or using small-scale HVAC systems. Online and real-time implementations of data-driven FDD methods in large-scale systems in real buildings, that can demonstrate the method's performances under various weather conditions, are still rare and in its infancy stage. Despite this limitation, some real-building deployments have been reported in the literature. Here real-building deployment refers to using a strategy for real-time FDD. For example, Lee et al. [116] deployed an SVM-based AI-FDD system to reject false alarms on 14 chillers in a data center for a year. Their results showed that the AI-FDD system achieved 100% correct false alarm rejection rate and the operational cost of investing in the AI platform can be recovered in three months. Wall and Guo [6] presented case studies using five different FDD tools in Australia, some of which reported to employ machine learning techniques. Blanes et al. [218] reported a CASCADE Implementation Kit that integrates FDD tested in two major European airports. The FDD tool employs the SVM method to detect sensor faults. Although the authors implemented pilot tests through 2013 to 2014, a complete performance evaluation was not found.

In general, reliable, fast and computationally affordable solutions that are readily deployable in the field have not been explored sufficiently [14]. Real-building deployments are challenging due to incomplete information and uncertainty [13]. The main factors contributing to this challenge are lack of sensors, poor sensor accuracy, imbalance of fault and fault-free training data, ad-hoc naming conventions for data points, non-standardized sensor installation and control logic, and missing data [13]. A recent study has shown that data uncertainty has a significant impact on the performance of SVM algorithms for chiller fault diagnosis [64]. Validation of data-driven FDD in terms of not only accuracy but also decision-making with real-world uncertainty is needed to overcome market barriers.

Performance evaluation, benchmarking, and fault impact analysis

In the literature, there are limited studies that compare the performance between FDD methods, especially under different categories (e.g., data-driven vs rule-based, supervised vs unsupervised). More comparison studies are needed to demonstrate the performance of and identify the weakness of data-driven methods. On the other hand, establishing common FDD datasets with validated ground-truth is needed to facilitate the assessment of different FDD methods. Lawrence Berkeley National Laboratory has released large FDD datasets, including experimental, simulated, and real building data, to support this effort [191].

In building HVAC systems, there are faults that generate relevant effects while others have negligible symptoms. However, the performance evaluation process of an FDD method is mostly based on the calculation of classification accuracy metrics without considering the importance of prioritizing faults with most adverse impacts. The impact assessment through novel weighted multi-criteria key performance indices (KPIs) is thus needed to put the right attention on different faults considering their effects in terms of energy consumption, greenhouse gas (GHG) emissions, energy costs, thermal comfort and indoor air quality (IAQ) according to their severity and occurrence frequency. For example, Chen et al. [219] developed a simulation-based framework for evaluating the fault effects in FCU. They also proposed a metric, namely the fault symptom occurrence probability (SOP), to assist the fault prioritization. Lu et al. [188] conducted a comprehensive fault impact analysis and robustness assessment of the high-performance control sequences from ASHRAE Guideline 36 using Modelica-based simulation with key performance indexes to evaluate fault impacts from the aspects of energy consumption and energy cost, control quality factor, thermal comfort, ventilation, and the power system.

Scalability and transferability

HVAC systems in large commercial buildings are typically designed and constructed in a unique way for each building. Each building may have its own unique boundary conditions, such as weather, occupancy, and internal load schedules that vary daily. As a result, data-driven FDD method developed for one building may not be applicable to another building. The following research areas may be considered to improve the scalability and transferability of a data-driven FDD method. **Hybrid approach.** Data-driven FDD methods require a considerable amount of data to exploit enough reliable and robust extracted knowledge. However, data-driven FDD methods generally cannot extrapolate well beyond the range of training data related to specific boundary conditions, limiting then the scalability and transferability of detection and diagnosis logics among different systems [220]. The expert-based approach, in contrast, has a strong capability for replicating and transferring expert diagnostic reasoning, especially in cases where initial information is not enough for deploying a data-driven process. Integration of both approaches may significantly improve the robustness, accuracy, and generalizability of FDD tools designed for building energy system applications.

Transfer learning. Besides the hybrid approach, transfer learning is being investigated as a fully data-driven solution to address the scalability issues of FDD strategies. Transfer learning [221], can effectively reduce the time to re-collect labeled data and re-train FDD algorithms, and thus reducing developmental costs. Recently, there has been some discussions about transfer learning in the building FDD field. For example, Dowling and Zhang [165] demonstrated a transferable Bayesian classifier for detecting supply fan degradation fault due to fouling filters in a VAV system. Miyata et al. [222] demonstrated transfer learning on convolutional neural networks (CNN) for fault diagnosis of central chilled water plants. Liu et al. [143] developed a transfer-learning-based CNNs for fault diagnosis of chillers. More studies are needed to further explore the potential of using transfer learning to improve the scalability of data-driven FDD.

Metadata schemas. Metadata schema or semantic data model allows data from different buildings to be described in a consistent and standardized manner. Using a common metadata schema not only eases the data collection process (as mentioned in Section 3.1), but also makes a data-driven FDD method more generic. Without a common metadata schema, a data-driven FDD method must be hard-coded to a specific data source of a specific building, thus limiting its scalability and transferability. In addition, metadata schemas can provide well-organized information about the nature of the data (e.g., type of sensor, causal relationship between points), which allows expert knowledge to be incorporated into a data-driven method more effectively. Project Haystack, Brick, and the recent ASHRAE 223P standard [22–24] are good options of metadata schema for building energy systems.

Interpretability

For a market-oriented FDD product, its interpretability (i.e., the ability to explain how a fault is detected or diagnosed) is very important. In fact, building professionals tend to be suspicious of the output of datadriven processes because they are unable to fully understand the model inference mechanism [223]. It is becoming more and more important to develop FDD tools that are capable of providing feedback about the reasons behind a certain detection or diagnosis result with robust indication of the supporting and conflicting evidences towards it.

In this respect, hybrid approaches, such as BNs that incorporate causal relationships between faults and symptoms, show great advantages. However, for pure black-box approaches, such as ANN, users are often unable to explain how they make decisions due to the models non-intuitive and non-transparent nature. The development of an explainable framework can help increase user confidence in such models. While interpretability continues to be a challenging task, a few studies have focused on this issue in recent years. For example, Madhikermi et al. [62] used Local Interpretable Model-agnostic Explanations (LIME) to explain behaviors of SVM and ANN in detecting AHU heat recycler fault. Li et al. [174] developed an explainable CNN-based FDD by utilizing Gradient-weighted Class Activation Mapping and validated it using the ASHRAE 1043-RP data.

Cyber security and data privacy

Modern BAS are typically connected to internet or enterprise network to reduce operational cost and increase automation. There are many benefits that a BAS can gain through the network connectivity, such as remote management, cloud computing, and data sharing [224, 225]. However, a network connectivity also makes BAS and the associated systems and devices potential targets of cyber attacks, leading to comprised systems and loss of credential information. For building energy systems, cyber attacks can disrupt the normal operation and result in serious consequences, such as occupant discomfort, energy waste, equipment downtime, and disruption of grid operation [225]. Therefore, there is a need for an FDD framework that takes cyber security and data privacy into account. For example, researchers are currently developing a Cyber Defense and Resilient System (CYDRES) that employs fault detection, fault diagnostics, fault prognosis, and cyber-resilient control scheme to enhance Grid-interactive Efficient Buildings (GEBs) tolerance to both cyber-related and physical faults [225,226].

Overall speaking, cyber security and data privacy is an important topic for any data-driven product in modern smart buildings. Further reading on this topic of can be found in Habibzadeh et al. [227].

User experience

Each of the previously mentioned future challenges will to some extend affect the user acceptance of data-driven FDD services. The user/client is usually the one paying for the service, thus creating a successful user experience will also benefit user acceptance of FDD as a service and ease a widespread implementation in real buildings. A positive user experience depends on how information is presented to the user to be able to understand what is happening in the building/system and why. In this regard, dedicated dashboards for different building users (facility managers, building owners, building tenants) are of utmost importance [228,229]. A proper visualization of measurement data, predicted data from data-driven models and case-specific key performance indicators all contribute to a better user experience and thus higher acceptance of data-driven methods among clients/users. A proper indication of the severity of a fault is important for the person, usually the facility manager, who has to repair the faulty systems. Clustering and ranking faults/alarms based on their severity and criticality for the operation of the system/building is a time-saving measure for the facility manager in an often hectic workday. More research is needed to understand how to improve the user experiences for data-driven FDD methods.

7. Conclusion

This paper provides a comprehensive review of the process, systems studied, and evaluation metrics for data-driven FDD. Existing literature provides promising methods and frameworks for implementing datadriven fault detection and fault diagnosis, step by step from collecting data to detecting anomaly to isolating root causes. Data-driven fault prognosis remains to be further developed. In terms of system studied, many studies exist that apply data-driven FDD methods to typical building HVAC systems (e.g., AHU-VAV and chiller). However, most of the studies are based on simulated or lab experiment data. Many types of evaluation metrics have been reported in the literature which are sufficient for data-driven FDD performance evaluation. Overall speaking, existing literature has laid a solid foundation to demonstrate the feasibility and benefit of using data-driven FDD. Yet significant challenges still remain for a wide market adoption of data-driven FDD methods. These challenges include real time and real-building implementation that is subject to data uncertainties; method performance benchmarking in real buildings; fault impact analysis; method scalability and transferability; fault interpretation; cyber security and data privacy; and user experience. It is our hope that this review would provide insights and directions for practitioners and researchers to develop the next generation data-driven FDD products.

Z. Chen et al.

Acronyms		LS-SVM	Least Squares Support Vector Machine	
AB	Adaptive Boosting		MultiLayer Perceptron	
AE	Autoencoder	NB	Naive Bayes	
AHU	Air Handling Unit	NN	Neural Network	
AI	Artificial Intelligence	OA	Outdoor Air	
ANN	Artificial Neural Network	PCA	Principal Component Analysis	
ADT	Adaptiva Personanga Theory	PLS	Partial Least Squares	
ARI		PLSR	Partial Least Square Regression	
AKX	Auto-Regressive model with exogenous variables	QDA	Quadratic Discriminant Analysis	
AUC	Area Under the Curve	RA	Return Air	
BAS	Building Automation System	RBF	Radial Basis Function network	
BIM	Building Information Model	RBM	Restricted Boltzmann Machine	
BN	Bayesian Network	RF	Random Forest	
CART	Classification And Regression Tree	RNN	Recurrent Neural Network	
CAV	Constant Air Volume	ROC	Receiver Operator Characteristic	
CNN	Convolutional Neural Network	RTU	RoofTop Unit	
СОР	Coefficient of Performance	RUL	Remaining Useful Life	
DBN	Deep Belief Network	SA	Supply Air	
DNN	Deen Neural Network	SGA	Simple Genetic Algorithm	
DRSCAN	Density-Based Spatial Clustering of Applications with Noise	SNN	Shallow Neural Network	
DT	Decision Tree	SVR	Support Vector Regression	
ECU		VAV	Variable Air Volume	
FCU		VRF	Variable Refrigerant Flow	
FDD	Fault Detection and Diagnostics	XGB	eXtreme Gradient Boosting	
GA	Genetic Algorithm	Doelanat	ion of comparing interest	
GAN	Generative Adversarial Network	Declarat		
GB	Gradient Boosting	The a cial inter	nuthors declare that they have no known competing finan- rests or personal relationships that could have appeared to	
GBM	Generalized Boosted regression Model	influence the work reported in this paper.		
GHG	GreenHouse Gas	Data availability No data was used for the research described in the article Acknowledgments		
HMM	Hidden Markov Model			
HVAC	Heating, Ventilation and Air Conditioning			
IAQ	Indoor Air Quality	This work has been performed within the framework of the Interna- tional Energy Agency - Energy in Buildings and Communities Program (IEA-EBC) Annex 81 "Data-Driven Smart Buildings". Drexel University and Texas A&M University's efforts were partly supported by: (1) NSF project # 2050509 "PFI-RP: Data-Driven Services for High Performance		
IGFF	Information Greedy Feature Filter			
KNN	K-Nearest Neighbors			
КРСА	Kernel Principal Components Analysis	and Sustainable Buildings." (2) the Building Technologies Office at the U.S. Department of Energy through the Emerging Technologies program under award number DE-EE0009150 and DE-EE0009153, and (2) the Building Program to the U.S. Department of Energy through the Emerging Technologies program under award number DE-EE0009150 and DE-EE0009153, and		
LAPART	Lateral Priming Adaptive Resonance Theory			
LDA	Linear Discriminant Analysis	(3) the Building America program at the U.S. Department of Energy under award number DE-EE0008694. Lawrence Berkeley National Lab-		
LinReg	Linear Regression	oratory's efforts were partly supported by the U.S. Department of Energy under contract number DE-AC0205CH11231. The views and		
LogReg	Logistic Regression	opinions of authors expressed herein do not necessarily state or reflection the second states Government or any agency thereof		

LogReg Logistic Regression

Z. Chen et al.

References

- Katipamula S, Brambley MR. Methods for fault detection, diagnostics, and prognostics for building systems—A review, Part I. HVAC&R Res 2005;11:3–25.
 Haves P. Overview of diagnostic methods. In: Proceedings of the workshop on
- diagnostics for commercial buildings: From research to practice. 1999, p. 16–7.
- [3] Kramer H, Lin G, Curtin C, Crowe E, Granderson J. Building analytics and monitoring-based commissioning: Industry practice, costs, and savings. Energy Effic 2020;13:537–49.
- [4] Hu G, Zhou T, Liu Q. Data-driven machine learning for fault detection and diagnosis in nuclear power plants: A review. Front Energy Res 2021;185.
- [5] Granderson J, Singla R, Mayhorn E, Ehrlich P, Vrabie D, Frank S. Characterization and survey of automated fault detection and diagnostic tools. Report Number LBNL-2001075, Lawrence Berkeley National Laboratory, 2017; 2017.
- [6] Wall J, Guo Y. Evaluation of next-generation automated fault detection & diagnostics (FDD) tools for commercial building energy efficiency. 2018, http: //dx.doi.org/10.25916/5ce9d6cec53af.
- [7] Katipamula S, Brambley MR. Methods for fault detection, diagnostics, and prognostics for building systems—A review, Part II. HVAC&R Res 2005;11:169–87.
- [8] Venkatasubramanian V, Rengaswamy R, Yin K, Kavuri SN. A review of process fault detection and diagnosis: Part I: Quantitative model-based methods. Comput Chem Eng 2003;27:293–311.
- [9] Venkatasubramanian V, Rengaswamy R, Kavuri SN. A review of process fault detection and diagnosis: Part II: Qualitative models and search strategies. Comput Chem Eng 2003;27:313–26.
- [10] Venkatasubramanian V, Rengaswamy R, Kavuri SN, Yin K. A review of process fault detection and diagnosis: Part III: Process history based methods. Comput Chem Eng 2003;27:327–46.
- [11] Rogers AP, Guo F, Rasmussen BP. A review of fault detection and diagnosis methods for residential air conditioning systems. Build Environ 2019;161:106236.
- [12] Chen J, Zhang L, Li Y, Shi Y, Gao X, Hu Y. A review of computingbased automated fault detection and diagnosis of heating, ventilation and air conditioning systems. Renew Sustain Energy Rev 2022;161:112395.
- [13] Zhao Y, Li T, Zhang X, Zhang C. Artificial intelligence-based fault detection and diagnosis methods for building energy systems: Advantages, challenges and the future. Renew Sustain Energy Rev 2019;109:85–101.
- [14] Mirnaghi MS, Haghighat F. Fault detection and diagnosis of large-scale HVAC systems in buildings using data-driven methods: A comprehensive review. Energy Build 2020;229:110492.
- [15] Nor NM, Hassan CRC, Hussain MA. A review of data-driven fault detection and diagnosis methods: Applications in chemical process systems. Rev Chem Eng 2020;36:513–53.
- [16] Zhang L, Wen J, Li Y, Chen J, Ye Y, Fu Y, et al. A review of machine learning in building load prediction. Appl Energy 2021;285:116452.
- [17] Zhang L, Chen Z, Zhang X, Pertzborn A, Jin X. Challenges and opportunities of machine learning control in building operations. Build Simul 2023. http: //dx.doi.org/10.1007/s12273-023-0984-6.
- [18] Chen J, Kher S, Somani A. Distributed fault detection of wireless sensor networks. In: Proceedings of the 2006 workshop on dependability issues in wireless ad hoc networks and sensor networks. DIWANS '06, New York, NY, USA: Association for Computing Machinery; 2006, p. 65–72.
- [19] Wang S, Xing J, Jiang Z, Dai Y. A novel sensors fault detection and selfcorrection method for HVAC systems using decentralized swarm intelligence algorithm. Int J Refrig 2019;106:54–65.
- [20] Bergés M, Lange H, Gao J. Data-driven operation of building systems: Present challenges and future prospects. In: Workshop of the European group for intelligent computing in engineering. Springer; 2018, p. 23–52.
- [21] Pradhan O, Hälleberg D, Chen Z, Wen J, Wu T, Candan KS, et al. Lagged-kNN based data imputation approach for multi-stream building systems data. Int High Perform Build Conf 2022.
- [22] Balaji B, Bhattacharya A, Fierro G, Gao J, Gluck J, Hong D, et al. Brick: Metadata schema for portable smart building applications. Appl Energy 2018;226:1273–92.
- [23] Bergmann H, Mosiman C, Saha A, Haile S, Livingood W, Bushby S, et al. Semantic interoperability to enable smart, grid-interactive efficient buildings. In: Summer study on energy efficiency in buildings 2020. ACEEE; 2020.
- [24] Pritoni M, Paine D, Fierro G, Mosiman C, Poplawski M, Saha A, et al. Metadata schemas and ontologies for building energy applications: A critical review and use case analysis. Energies 2021;14. http://dx.doi.org/10.3390/en14072024, URL https://www.mdpi.com/1996-1073/14/7/2024.
- [25] Ploennigs J, Maghella M, Schumann A, Chen B. Semantic diagnosis approach for buildings. IEEE Trans Ind Inf 2017;13:3399–410. http://dx.doi.org/10.1109/ TII.2017.2726001.
- [26] Pritoni M, Weyandt C, Carter D, Elliott J. Towards a scalable model for smart buildings. In: Summer study on energy efficiency in buildings 2018. ACEEE; 2018.
- [27] Jin Z, Yezheng W, Gang Y. A stochastic method to generate bin weather data in Nanjing, China. Energy Convers Manage 2006;47:1843–50.

- [28] Ouyang T, Zha X, Qin L. A combined multivariate model for wind power prediction. Energy Convers Manage 2017;144:361–73.
- [29] Kasam AA, Lee BD, Paredis CJJ. Statistical methods for interpolating missing meteorological data for use in building simulation. Build Simul 2014;7:455–65.
- [30] Garnier A, Eynard J, Caussanel M, Grieu S. Missing data estimation for energy resources management in tertiary buildings. In: CCCA12. IEEE; 2012, p. 1–6.
- [31] Rahman A, Srikumar V, Smith AD. Predicting electricity consumption for commercial and residential buildings using deep recurrent neural networks. Appl Energy 2018;212:372–85.
- [32] Yang J, Tan KK, Santamouris M, Lee SE. Building energy consumption raw data forecasting using data cleaning and deep recurrent neural networks. Buildings 2019;9:204.
- [33] Inman D, Elmore R, Bush B. A case study to examine the imputation of missing data to improve clustering analysis of building electrical demand. Build Serv Eng Res Technol 2015;36:628–37.
- [34] Zhang L. A pattern-recognition-based ensemble data imputation framework for sensors from building energy systems. Sensors 2020;20:5947.
- [35] Li D, Zhou Y, Hu G, Spanos CJ. Handling incomplete sensor measurements in fault detection and diagnosis for building HVAC systems. IEEE Trans Autom Sci Eng 2019;17:833–46.
- [36] Wang Z, Wang L, Tan Y, Yuan J. Fault detection based on Bayesian network and missing data imputation for building energy systems. Appl Therm Eng 2021;182:116051.
- [37] Fan C, Chen M, Wang X, Wang J, Huang B. A review on data preprocessing techniques toward efficient and reliable knowledge discovery from building operational data. Front Energy Res 2021;9:652801.
- [38] Zhang L, Frank S, Kim J, Jin X, Leach M. A systematic feature extraction and selection framework for data-driven whole-building automated fault detection and diagnostics in commercial buildings. Build Environ 2020;186:107338.
- [39] Zhang L, Wen J. A systematic feature selection procedure for short-term data-driven building energy forecasting model development. Energy Build 2019;183:428–42.
- [40] Chandrashekar G, Sahin F. A survey on feature selection methods. Comput Electr Eng 2014;40:16–28.
- [41] Li D, Zhou Y, Hu G, Spanos CJ. Optimal sensor configuration and feature selection for AHU fault detection and diagnosis. IEEE Trans Ind Inf 2016;13:1369–80.
- [42] Li S, Wen J. Development and validation of a dynamic air handling unit model, part I. ASHRAE Trans 2010;116:45.
- [43] Li S, Wen J, Zhou X, Klaassen CJ. Development and validation of a dynamic air handling unit model, part 2. ASHRAE Trans 2010;116:57–73.
- [44] Namburu SM, Azam MS, Luo J, Choi K, Pattipati KR. Data-driven modeling, fault diagnosis and optimal sensor selection for HVAC chillers. IEEE Trans Autom Sci Eng 2007;4:469–73.
- [45] Yan K, Shen W, Mulumba T, Afshari A. ARX model based fault detection and diagnosis for chillers using support vector machines. Energy Build 2014;81:287–95.
- [46] Mulumba T, Afshari A, Yan K, Shen W, Norford LK. Robust model-based fault diagnosis for air handling units. Energy Build 2015;86:698–707. http://dx. doi.org/10.1016/j.enbuild.2014.10.069, URL https://www.sciencedirect.com/ science/article/pii/S0378778814009153.
- [47] Chen Y, Wen J, Lo J. Using weather and schedule based pattern matching and feature based principal component analysis for whole building fault detection— Part II field evaluation. ASME J Eng Sustain Build Cities 2021;3. http://dx.doi. org/10.1115/1.4052730, 011002.
- [48] Chen Y, Wen J, Lo J. Using weather and schedule-based pattern matching and feature-based principal component analysis for whole building fault detection–Part I development of the method. ASME J Eng Sustain Build Cities 2022;3.
- [49] Yan R, Ma Z, Zhao Y, Kokogiannakis G. A decision tree based data-driven diagnostic strategy for air handling units. Energy Build 2016;133:37–45.
- [50] Han H, Gu B, Hong Y, Kang J. Automated FDD of multiple-simultaneous faults (MSF) and the application to building chillers. Energy Build 2011;43:2524–32.
- [51] Han H, Gu B, Kang J, Li ZR. Study on a hybrid SVM model for chiller FDD applications. Appl Therm Eng 2011;31:582–92.
- [52] Miyata S, Lim J, Akashi Y, Kuwahara Y, Tanaka K. Fault detection and diagnosis for heat source system using convolutional neural network with imaged faulty behavior data. Sci Technol Built Environ 2020;26:52–60.
- [53] Lin G, Claridge DE. A temperature-based approach to detect abnormal building energy consumption. Energy Build 2015;93:110–8.
- [54] Miller C, Nagy Z, Schlueter A. Automated daily pattern filtering of measured building performance data. Autom Constr 2015;49:1–17.
- [55] Fan C, Xiao F, Madsen H, Wang D. Temporal knowledge discovery in big BAS data for building energy management. Energy Build 2015;109:75–89.
- [56] Piscitelli MS, Mazzarelli DM, Capozzoli A. Enhancing operational performance of AHUs through an advanced fault detection and diagnosis process based on temporal association and decision rules. Energy Build 2020;226:110369. http://dx.doi.org/10.1016/j.enbuild.2020.110369.
- [57] Li X, Bowers CP, Schnier T. Classification of energy consumption in buildings with outlier detection. IEEE Trans Ind Electron 2009;57:3639–44.

- [58] Huang J, Wu T, Yoon H, Pradhan O, Wen J, O'Neill Z. Automatic fault detection baseline construction for building HVAC systems using joint entropy and enthalpy. In: IIE annual conference. Proceedings. Institute of Industrial and Systems Engineers (IISE); 2021, p. 536–41.
- [59] Liang J, Du R. Model-based fault detection and diagnosis of HVAC systems using support vector machine method. Int J Refrig 2007;30:1104–14.
- [60] Dehestani D, Su S, Nguyen H, Guo Y. Robust fault tolerant application for HVAC system based on combination of online SVM and ANN black box model. In: 2013 European control conference. ECC, IEEE; 2013, p. 2976–81.
- [61] Beghi A, Cecchinato L, Corazzol C, Rampazzo M, Simmini F, Susto GA. A oneclass SVM based tool for machine learning novelty detection in HVAC chiller systems. IFAC Proc Vol 2014;47:1953–8.
- [62] Madhikermi M, Malhi AK, Främling K. Explainable artificial intelligence based heat recycler fault detection in air handling unit. In: International workshop on explainable, transparent autonomous agents and multi-agent systems. Springer; 2019, p. 110–25.
- [63] Montazeri A, Kargar SM. Fault detection and diagnosis in air handling using data-driven methods. J Build Eng 2020;31:101388.
- [64] Li X, Liu J, Liu B, Zhang Q, Li K, Dong Z, et al. Impacts of data uncertainty on the performance of data-driven-based building fault diagnosis. J Build Eng 2021;43:103153.
- [65] Capozzoli A, Lauro F, Khan I. Fault detection analysis using data mining techniques for a cluster of smart office buildings. Expert Syst Appl 2015;42:4324–38. http://dx.doi.org/10.1016/j.eswa.2015.01.010, URL https://www.sciencedirect. com/science/article/pii/S0957417415000251.
- [66] Capozzoli A, Piscitelli MS, Brandi S, Grassi D, Chicco G. Automated load pattern learning and anomaly detection for enhancing energy management in smart buildings. Energy 2018;157:336–52.
- [67] Tesfay M, Alsaleem F, Sinkar K, Arunasalam P. Superheat prediction & fault diagnostics of HVAC from simple temperature measurements using big data approach. 2021.
- [68] Liu X, Ding Y, Tang H, Xiao F. A data mining-based framework for the identification of daily electricity usage patterns and anomaly detection in building electricity consumption data. Energy Build 2021;231:110601.
- [69] Piscitelli MS, Brandi S, Capozzoli A, Xiao F. A data analytics-based tool for the detection and diagnosis of anomalous daily energy patterns in buildings. Build Simul 2021;14:131–47.
- [70] Chiosa R, Piscitelli MS, Capozzoli A. A data analytics-based energy information system (EIS) tool to perform meter-level anomaly detection and diagnosis in buildings. Energies 2021;14. http://dx.doi.org/10.3390/en14010237, URL https://www.mdpi.com/1996-1073/14/1/237.
- [71] Chiosa R, Piscitelli MS, Fan C, Capozzoli A. Towards a self-tuned data analytics-based process for an automatic context-aware detection and diagnosis of anomalies in building energy consumption timeseries. Energy Build 2022;270:112302. http://dx.doi.org/10.1016/j.enbuild.2022.112302.
- [72] Zhao Y, Wang S, Xiao F. A system-level incipient fault-detection method for HVAC systems. HVAC&R Res 2013;19:593–601.
- [73] Liu J, Liu J, Chen H, Yuan Y, Li Z, Huang R. Energy diagnosis of variable refrigerant flow (VRF) systems: Data mining technique and statistical quality control approach. Energy Build 2018;175:148–62.
- [74] Magoulès F, Zhao H-x, Elizondo D. Development of an RDP neural network for building energy consumption fault detection and diagnosis. Energy Build 2013;62:133–8.
- [75] Du Z, Fan B, Jin X, Chi J. Fault detection and diagnosis for buildings and HVAC systems using combined neural networks and subtractive clustering analysis. Build Environ 2014;73:1–11.
- [76] Jones CB. Fault detection and diagnostics of an HVAC sub-system using adaptive resonance theory neural networks. The University of New Mexico; 2015.
- [77] Lee K-P, Wu B-H, Peng S-L. Deep-learning-based fault detection and diagnosis of air-handling units. Build Environ 2019;157:24–33.
- [78] Sipple J. Interpretable, multidimensional, multimodal anomaly detection with negative sampling for detection of device failure. In: International conference on machine learning. PMLR; 2020, p. 9016–25.
- [79] Xu C, Chen H. A hybrid data mining approach for anomaly detection and evaluation in residential buildings energy data. Energy Build 2020;215:109864.
- [80] Asgari S, Gupta R, Puri IK, Zheng R. A data-driven approach to simultaneous fault detection and diagnosis in data centers. Appl Soft Comput 2021;110:107638.
- [81] Liao H, Cai W, Cheng F, Dubey S, Rajesh PB. An online data-driven fault diagnosis method for air handling units by rule and convolutional neural networks. Sensors 2021;21:4358.
- [82] Yan K, Chong A, Mo Y. Generative adversarial network for fault detection diagnosis of chillers. Build Environ 2020;172:106698.
- [83] Fan C, Xiao F, Yan C. A framework for knowledge discovery in massive building automation data and its application in building diagnostics. Autom Constr 2015;50:81–90.
- [84] Cheng Z, Zhao Q, Wang F, Chen Z, Jiang Y, Li Y. Case studies of fault diagnosis and energy saving in buildings using data mining techniques. In: 2016 IEEE international conference on automation science and engineering. CASE, IEEE; 2016, p. 646–51.

- [85] Li D, Hu G, Spanos CJ. A data-driven strategy for detection and diagnosis of building chiller faults using linear discriminant analysis. Energy Build 2016;128:519–29.
- [86] Dey M, Rana SP, Dudley S. A case study based approach for remote fault detection using multi-level machine learning in a smart building. Smart Cities 2020;3:401–19.
- [87] Dey M, Rana SP, Dudley S. Smart building creation in large scale HVAC environments through automated fault detection and diagnosis. Future Gener Comput Syst 2020;108:950–66.
- [88] Aguilar J, Ardila D, Avendaño A, Macias F, White C, Gomez-Pulido J, et al. An autonomic cycle of data analysis tasks for the supervision of HVAC systems of smart building. Energies 2020;13:3103.
- [89] Gunay HB, Shi Z. Cluster analysis-based anomaly detection in building automation systems. Energy Build 2020;228:110445.
- [90] Xu Y, Yan C, Shi J, Lu Z, Niu X, Jiang Y, et al. An anomaly detection and dynamic energy performance evaluation method for HVAC systems based on data mining. Sustain Energy Technol Assess 2021;44:101092.
- [91] Zhou X, Yang T, Liang L, Zi X, Yan J, Pan D. Anomaly detection method of daily energy consumption patterns for central air conditioning systems. J Build Eng 2021;38:102179.
- [92] Wang S, Qin J. Sensor fault detection and validation of VAV terminals in air conditioning systems. Energy Convers Manage 2005;46:2482–500.
- [93] Du Z, Jin X. Detection and diagnosis for sensor fault in HVAC systems. Energy Convers Manage 2007;48:693–702.
- [94] Du Z, Jin X. Multiple faults diagnosis for sensors in air handling unit using Fisher discriminant analysis. Energy Convers Manage 2008;49:3654–65.
- [95] Hu Y, Chen H, Xie J, Yang X, Zhou C. Chiller sensor fault detection using a self-adaptive principal component analysis method. Energy Build 2012;54:252–8.
- [96] Li S, Wen J. Application of pattern matching method for detecting faults in air handling unit system. Autom Constr 2014;43:49–58.
- [97] Li S, Wen J. A model-based fault detection and diagnostic methodology based on PCA method and wavelet transform. Energy Build 2014;68:63–71.
- [98] Cotrufo N, Zmeureanu R. PCA-based method of soft fault detection and identification for the ongoing commissioning of chillers. Energy Build 2016;130:443–52.
- [99] Hu Y, Chen H, Li G, Li H, Xu R, Li J. A statistical training data cleaning strategy for the PCA-based chiller sensor fault detection, diagnosis and data reconstruction method. Energy Build 2016;112:270–8.
- [100] Shi S, Li G, Chen H, Hu Y, Wang X, Guo Y, et al. An efficient VRF system fault diagnosis strategy for refrigerant charge amount based on PCA and dual neural network model. Appl Therm Eng 2018;129:1252–62.
- [101] Li G, Hu Y. Improved sensor fault detection, diagnosis and estimation for screw chillers using density-based clustering and principal component analysis. Energy Build 2018;173:502–15.
- [102] Zhou Z, Chen H, Li G, Zhong H, Zhang M, Wu J. Data-driven fault diagnosis for residential variable refrigerant flow system on imbalanced data environments. Int J Refrig 2021;125:34–43.
- [103] Lampis M, Andrews JD. Bayesian belief networks for system fault diagnostics. Qual Reliab Eng Int 2009;25:409–26.
- [104] Zhao Y, Wen J, Xiao F, Yang X, Wang S. Diagnostic Bayesian networks for diagnosing air handling units faults-Part I: Faults in dampers, fans, filters and sensors. Appl Therm Eng 2017;111:1272–86.
- [105] Zhao Y, Wen J, Wang S. Diagnostic Bayesian networks for diagnosing air handling units faults-Part II: Faults in coils and sensors. Appl Therm Eng 2015;90:145–57.
- [106] Xiao F, Zhao Y, Wen J, Wang S. Bayesian network based FDD strategy for variable air volume terminals. Autom Constr 2014;41:106–18.
- [107] Verbert K, Babuška R, De Schutter B. Combining knowledge and historical data for system-level fault diagnosis of HVAC systems. Eng Appl Artif Intell 2017;59:260–73.
- [108] Chen Y, Wen J, Pradhan O, Lo J, Wu T. Using discrete Bayesian networks for diagnosing and isolating cross-level faults in HVAC systems. Appl Energy 2022.
- [109] İçen D, Ersel D. A new approach for probability calculation of fuzzy events in Bayesian networks. Internat J Approx Reason 2019;108:76–88.
- [110] Rohmer J. Uncertainties in conditional probability tables of discrete Bayesian belief networks: A comprehensive review. Eng Appl Artif Intell 2020;88:103384.
- [111] Shi Z, O'Brien W, Gunay HB. Development of a distributed building fault detection, diagnostic, and evaluation system. ASHRAE Trans 2018;124:23–37.
- [112] Zio E, Peloni G. Particle filtering prognostic estimation of the remaining useful life of nonlinear components. Reliab Eng Syst Saf 2011;96:403–9.
- [113] Li X, Shao M, Ding L. HVAC fault diagnosis system using rough set theory and support vector machine. In: 2009 second international workshop on knowledge discovery and data mining. IEEE; 2009, p. 895–9.
- [114] Beghi A, Brignoli R, Cecchinato L, Menegazzo G, Rampazzo M. A data-driven approach for fault diagnosis in HVAC chiller systems. In: 2015 IEEE conference on control applications. CCA, IEEE; 2015, p. 966–71.
- [115] Fan Y, Cui X, Han H, Lu H. Chiller fault diagnosis with field sensors using the technology of imbalanced data. Appl Therm Eng 2019;159:113933. http://dx.doi.org/10.1016/j.applthermaleng.2019.113933, URL https://www. sciencedirect.com/science/article/pii/S1359431119304314.

- [116] Lee D, Lai C-W, Liao K-K, Chang J-W. Artificial intelligence assisted false alarm detection and diagnosis system development for reducing maintenance cost of chillers at the data centre. J Build Eng 2021;36:102110.
- [117] Jin B, Li D, Srinivasan S, Ng S-K, Poolla K, Sangiovanni-Vincentelli A. Detecting and diagnosing incipient building faults using uncertainty information from deep neural networks. In: 2019 IEEE international conference on prognostics and health management. ICPHM, IEEE; 2019, p. 1–8.
- [118] Li T, Zhao Y, Zhang C, Luo J, Zhang X. A knowledge-guided and datadriven method for building HVAC systems fault diagnosis. Build Environ 2021;198:107850.
- [119] Zhong K, Han M, Han B. Data-driven based fault prognosis for industrial systems: A concise overview. IEEE/CAA J Autom Sin 2019;7:330–45.
- [120] Yu Y, Woradechjumroen D, Yu D. A review of fault detection and diagnosis methodologies on air-handling units. Energy Build 2014;82:550–62.
- [121] Davis J M, Martinkauppi I, Witte H, Berglöf K, Vallin S. IEA HPT Annex 52-longterm performance monitoring of GSHP systems for commercial, institutional and multi-family buildings: Guidelines for instrumentation and data. 2021, http://dx.doi.org/10.23697/tgr4-qn89.
- [122] Gehlin S, Spitler JD, Witte H, Andersson O, Berglöf K, Davis M, et al. IEA HPT Annex 52-long-term performance monitoring of GSHP systems for commercial, institutional and multifamily buildings: Guide for analysis and reporting of GSHP system performance — system boundaries and key performance indicators (KPI). 2022, http://dx.doi.org/10.23697/xa7z-vd92.
- [123] Yang C, Shen W, Chen Q, Gunay B. A practical solution for HVAC prognostics: Failure mode and effects analysis in building maintenance. J Build Eng 2018;15:26–32.
- [124] Yang C, Gunay B, Shi Z, Shen W. Machine learning-based prognostics for central heating and cooling plant equipment health monitoring. IEEE Trans Autom Sci Eng 2020;18:346–55.
- [125] Ahmad N, Atta R. Cost-effective wireless-controlled motor failure prediction for HVAC system in large buildings using demodulated current signature analysis. Life Sci J 2014;11.
- [126] Wang P, Gao RX, Fan Z. Switching local search particle filtering for heat exchanger degradation prognosis. In: 2015 IEEE international instrumentation and measurement technology conference proceedings. IEEE; 2015, p. 539–44.
- [127] Yan Y, Luh PB, Pattipati KR. Fault diagnosis of HVAC air-handling systems considering fault propagation impacts among components. IEEE Trans Autom Sci Eng 2017;14:705–17.
- [128] Arunthavanathan R, Khan F, Ahmed S, Imtiaz S. A deep learning model for process fault prognosis. Process Saf Environ Prot 2021;154:467–79.
- [129] Liu C, Yao R, Zhang L, Liao Y. Attention based echo state network: A novel approach for fault prognosis. In: Proceedings of the 2019 11th international conference on machine learning and computing. 2019, p. 489–93.
- [130] Wu Q, Ding K, Huang B. Approach for fault prognosis using recurrent neural network. J Intell Manuf 2020;31:1621–33.
- [131] He Z, Shao H, Ding Z, Jiang H, Cheng J. Modified deep autoencoder driven by multisource parameters for fault transfer prognosis of aeroengine. IEEE Trans Ind Electron 2021;69:845–55.
- [132] Niu G, Wang X, Liu E, Zhang B. Lebesgue sampling based deep belief network for lithium-ion battery diagnosis and prognosis. IEEE Trans Ind Electron 2021;69:8481–90.
- [133] Amruthnath N, Gupta T. A research study on unsupervised machine learning algorithms for early fault detection in predictive maintenance. In: 2018 5th International conference on industrial engineering and applications. ICIEA, IEEE; 2018, p. 355–61.
- [134] Narayanaswamy B, Balaji B, Gupta R, Agarwal Y. Data driven investigation of faults in HVAC systems with model, cluster and compare (MCC). In: Proceedings of the 1st ACM conference on embedded systems for energy-efficient buildings. 2014, p. 50–9.
- [135] Li G, Chen H, Hu Y, Wang J, Guo Y, Liu J, et al. An improved decision treebased fault diagnosis method for practical variable refrigerant flow system using virtual sensor-based fault indicators. Appl Therm Eng 2018;129:1292–303.
- [136] Yan R, Ma Z, Kokogiannakis G, Zhao Y. A sensor fault detection strategy for air handling units using cluster analysis. Autom Constr 2016;70:77–88.
- [137] Cho S-H, Hong Y-J, Kim W-T, Zaheer-Uddin M. Multi-fault detection and diagnosis of HVAC systems: An experimental study. Int J Energy Res 2005;29:471–83.
- [138] Hou Z, Lian Z, Yao Y, Yuan X. Data mining based sensor fault diagnosis and validation for building air conditioning system. Energy Convers Manage 2006;47:2479–90.
- [139] Du Z, Jin X, Yang Y. Fault diagnosis for temperature, flow rate and pressure sensors in VAV systems using wavelet neural network. Appl Energy 2009;86:1624–31.
- [140] Guo Y, Li G, Chen H, Wang J, Guo M, Sun S, et al. Optimized neural networkbased fault diagnosis strategy for VRF system in heating mode using data mining. Appl Therm Eng 2017;125:1402–13.
- [141] Shahnazari H, Mhaskar P, House JM, Salsbury TI. Modeling and fault diagnosis design for HVAC systems using recurrent neural networks. Comput Chem Eng 2019;126:189–203.

- [142] Han H, Xu L, Cui X, Fan Y. Novel chiller fault diagnosis using deep neural network (DNN) with simulated annealing (SA). Int J Refrig 2021;121:269–78.
- [143] Liu J, Zhang Q, Li X, Li G, Liu Z, Xie Y, et al. Transfer learning-based strategies for fault diagnosis in building energy systems. Energy Build 2021;250:111256.
- [144] Taheri S, Ahmadi A, Mohammadi-Ivatloo B, Asadi S. Fault detection diagnostic for HVAC systems via deep learning algorithms. Energy Build 2021;250:111275.
- [145] Wall J, Guo Y, Li J, West S. A dynamic machine learning-based technique for automated fault detection in HVAC systems. ASHRAE Trans 2011;117.
 [146] Zhao Y, Xiao F, Wang S. An intelligent chiller fault detection and diagnosis
- [140] Zhao F, Wang S. An Internet Chiner rate detection and dragnosis methodology using Bayesian belief network. Energy Build 2013;57:278–88.
 [147] Dong B. O'Neill Z. Li Z. A BIM-enabled information infrastructure for building
- energy fault detection and diagnostics. Autom Constr 2014;44:197–211. [148] Taal A, Itard L, Zeiler W. A reference architecture for the integration of
- [140] Tati Y, Jian L, Zcher W. A Terefore architecture for the megation of automated energy performance fault diagnosis into HVAC systems. Energy Build 2018;179:144–55. http://dx.doi.org/10.1016/j.enbuild.2018.08.031, URL https://www.sciencedirect.com/science/article/pii/S0378778818310570.
- [149] Wang Y, Wang Z, He S, Wang Z. A practical chiller fault diagnosis method based on discrete Bayesian network. Int J Refrig 2019;102:159–67.
- [150] West SR, Guo Y, Wang XR, Wall J. Automated fault detection and diagnosis of HVAC subsystems using statistical machine learning. In: 12th international conference of the international building performance simulation association. 2011, p. 2659–65.
- [151] Yan Y, Luh PB, Pattipati KR. Fault prognosis of key components in HVAC airhandling systems at component and system levels. IEEE Trans Autom Sci Eng 2020;17:2145–53.
- [152] Zhong C, Yan K, Dai Y, Jin N, Lou B. Energy efficiency solutions for buildings: Automated fault diagnosis of air handling units using generative adversarial networks. Energies 2019;12:527.
- [153] Yan K, Huang J, Shen W, Ji Z. Unsupervised learning for fault detection and diagnosis of air handling units. Energy Build 2020;210:109689.
- [154] Li D, Cheng F, Cai H, Zhang X, Cai W. A semi-supervised approach to fault detection and diagnosis for building HVAC systems based on the modified generative adversarial network. Energy Build 2021;246:111044.
- [155] Araya DB, Grolinger K, ElYamany HF, Capretz MAM, Bitsuamlak G. An ensemble learning framework for anomaly detection in building energy consumption. Energy Build 2017;144:191–206.
- [156] Fan C, Xiao F, Zhao Y, Wang J. Analytical investigation of autoencoder-based methods for unsupervised anomaly detection in building energy data. Appl Energy 2018;211:1123–35.
- [157] Han H, Zhang Z, Cui X, Meng Q. Ensemble learning with member optimization for fault diagnosis of a building energy system. Energy Build 2020;226:110351.
- [158] Li D, Zhou Y, Hu G, Spanos CJ. Fault detection and diagnosis for building cooling system with a tree-structured learning method. Energy Build 2016;127:540–51.
- [159] Guo Y, Tan Z, Chen H, Li G, Wang J, Huang R, et al. Deep learning-based fault diagnosis of variable refrigerant flow air-conditioning system for building energy saving. Appl Energy 2018;225:732–45.
- [160] Zhou Z, Li G, Wang J, Chen H, Zhong H, Cao Z. A comparison study of basic data-driven fault diagnosis methods for variable refrigerant flow system. Energy Build 2020;224:110232.
- [161] Ebrahimifakhar A, Kabirikopaei A, Yuill D. Data-driven fault detection and diagnosis for packaged rooftop units using statistical machine learning classification methods. Energy Build 2020;225:110318.
- [162] Shohet R, Kandil MS, Wang Y, McArthur JJ. Fault detection for non-condensing boilers using simulated building automation system sensor data. Adv Eng Inform 2020;46:101176.
- [163] Li B, Cheng F, Zhang X, Cui C, Cai W. A novel semi-supervised datadriven method for chiller fault diagnosis with unlabeled data. Appl Energy 2021;285:116459.
- [164] Wang Z, Wang L, Tan Y, Yuan J, Li X. Fault diagnosis using fused reference model and Bayesian network for building energy systems. J Build Eng 2021;34:101957.
- [165] Dowling CP, Zhang B. Transfer learning for HVAC system fault detection. In: 2020 American control conference. ACC, IEEE; 2020, p. 3879–85.
- [166] Turner WJN, Staino A, Basu B. Residential HVAC fault detection using a system identification approach. Energy Build 2017;151:1–17.
- [167] Choi K, Namburu SM, Azam MS, Luo J, Pattipati KR, Patterson-Hine A. Fault diagnosis in HVAC chillers. IEEE Instrum Meas Mag 2005;8:24–32.
- [168] Namburu SM, Luo J, Azam M, Choi K, Pattipati KR. Fault detection, diagnosis, and data-driven modeling in HVAC chillers. In: Signal processing, sensor fusion, and target recognition XIV, vol. 5809. SPIE; 2005, p. 143–54.
- [169] Zhao Y, Xiao F, Wen J, Lu Y, Wang S. A robust pattern recognitionbased fault detection and diagnosis (FDD) method for chillers. HVAC&R Res 2014;20:798–809.
- [170] Beghi A, Cecchinato L, Peterle F, Rampazzo M, Simmini F. Model-based fault detection and diagnosis for centrifugal chillers. In: 2016 3rd Conference on control and fault-tolerant systems. SysTol, IEEE; 2016, p. 158–63.
- [171] van de Sand R, Falk C, Corasaniti S, Reiff-Stephan J. A data-driven fault diagnosis approach towards oil retention in vapour compression refrigeration systems. In: 2019 international IEEE conference and workshop in Óbuda on electrical and power engineering. CANDO-EPE, IEEE; 2019, p. 197–202.

- [172] Liu J, Shi D, Li G, Xie Y, Li K, Liu B, et al. Data-driven and association rule mining-based fault diagnosis and action mechanism analysis for building chillers. Energy Build 2020;216:109957.
- [173] Yan K. Chiller fault detection and diagnosis with anomaly detective generative adversarial network. Build Environ 2021;201:107982.
- [174] Li G, Yao Q, Fan C, Zhou C, Wu G, Zhou Z, et al. An explainable onedimensional convolutional neural networks based fault diagnosis method for building heating, ventilation and air conditioning systems. Build Environ 2021;203:108057.
- [175] Gao Y, Han H, Ren ZX, Gao JQ, Jiang SX, Yang YT. Comprehensive study on sensitive parameters for chiller fault diagnosis. Energy Build 2021;251:111318.
- [176] Zucker G, Malinao J, Habib U, Leber T, Preisler A, Judex F. Improving energy efficiency of buildings using data mining technologies. In: 2014 IEEE 23rd international symposium on industrial electronics. ISIE, IEEE; 2014, p. 2664–9.
- [177] Li G, Hu Y, Chen H, Li H, Hu M, Guo Y, et al. A sensor fault detection and diagnosis strategy for screw chiller system using support vector data description-based D-statistic and DV-contribution plots. Energy Build 2016;133:230–45.
- [178] Liu J, Li G, Liu B, Li K, Chen H. Knowledge discovery of data-driven-based fault diagnostics for building energy systems: A case study of the building variable refrigerant flow system. Energy 2019;174:873–85.
- [179] Zeng Y, Chen H, Xu C, Cheng Y, Gong Q. A hybrid deep forest approach for outlier detection and fault diagnosis of variable refrigerant flow system. Int J Refrig 2020;120:104–18.
- [180] Zogg D, Shafai E, Geering HP. Fault diagnosis for heat pumps with parameter identification and clustering. Control Eng Pract 2006;14:1435–44.
- [181] Xue P, Zhou Z, Fang X, Chen X, Liu L, Liu Y, et al. Fault detection and operation optimization in district heating substations based on data mining techniques. Appl Energy 2017;205:926–40.
- [182] Jacob D, Dietz S, Komhard S, Neumann C, Herkel S. Black-box models for fault detection and performance monitoring of buildings. J Build Perform Simul 2010;3:53–62.
- [183] Wen J, Chen Y, Regnier A. Building fault detection and diagnostics. In: Baillieul J, Samad T, editors. Encyclopedia of systems and control. London: Springer London; 2019, p. 1–6. http://dx.doi.org/10.1007/978-1-4471-5102-9_100080-1.
- [184] Wang S, Chen Y. Fault-tolerant control for outdoor ventilation air flow rate in buildings based on neural network. Build Environ 2002;37:691–704.
- [185] Zhang L, Leach M, Bae Y, Cui B, Bhattacharya S, Lee S, et al. Sensor impact evaluation and verification for fault detection and diagnostics in building energy systems: A review. Adv Appl Energy 2021;3:100055.
- [186] Weimer J, Ahmadi SA, Araujo J, Mele FM, Papale D, Shames I, et al. Active actuator fault detection and diagnostics in HVAC systems. In: Proceedings of the fourth ACM workshop on embedded sensing systems for energy-efficiency in buildings. 2012, p. 107–14.
- [187] Torabi N, Gunay HB, O'Brien W, Barton T. Common human errors in design, installation, and operation of VAV AHU control systems—A review and a practitioner interview. Build Environ 2022;109333.
- [188] Lu X, Fu Y, O'Neill Z, Wen J. A holistic fault impact analysis of the highperformance sequences of operation for HVAC systems: Modelica-based case study in a medium-office building. Energy Build 2021;252:111448.
- [189] Hu Y, Yuill DP, Rooholghodos SA, Ebrahimifakhar A, Chen Y. Impacts of simultaneous operating faults on cooling performance of a high efficiency residential heat pump. Energy Build 2021;242:110975.
- [190] Hu Y, Yuill DP. Multiple simultaneous faults' impacts on air-conditioner behavior and performance of a charge diagnostic method. Appl Therm Eng 2022;215:119015.
- [191] Granderson J, Lin G, Chen Y, Casillas A, Wen J, Chen Z, et al. LBNL fault detection and diagnostics datasets. 2022, URL https://faultdetection.lbl.gov/ data/.
- [192] Wen J, Li S. RP-1312–Tools for evaluating fault detection and diagnostic methods for air-handling units. Tech Rep, ASHRAE; 2012.
- [193] Comstock MC, Braun JE. Development of analysis tools for the evaluation of fault detection and diagnostics in chillers. ASHRAE research project RP-1043, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Inc.; 1999.
- [194] Granderson J, Lin G, Harding A, Im P, Chen Y. Building fault detection data to aid diagnostic algorithm creation and performance testing. Sci Data 2020;7:1–14.
- [195] Li Y, O'Neill Z. A critical review of fault modeling of HVAC systems in buildings. Build Simul 2018;11:953–75.
- [196] Chen Z. Advanced solver development for large-scale dynamic building system simulation. Drexel University; 2019.
- [197] Chen Z, Wen J, Kearsley A, Pertzborn A. Smoothing techniques in dynamic building system simulation. In: 2021 international conference on instrumentation, control, and automation. ICA, IEEE; 2021, p. 156–61.
- [198] Chen Z, Wen J, Kearsley AJ, Pertzborn A. Evaluating the performance of an Inexact Newton method with a preconditioner for dynamic building system simulation. J Build Perform Simul 2022;15:112–27.

- [199] Granderson J, Lin G, Chen Y, Casillas A, Im P, Jung S, et al. LBNL fault detection and diagnostics datasets. 2022, http://dx.doi.org/10.25984/1881324, URL https://www.osti.gov/biblio/1881324.
- [200] Granderson J, Lin G, Chen Y, Casillas A, Wen J, Chen Z. LBNL fault detection and diagnostics data sets: Fan power unit. 2022.
- [201] DrivenData. Power laws: Detecting anomalies in usage. URL https://www. drivendata.org/competitions/52/anomaly-detection-electricity/page/102/.
- [202] Wang L, Haves P. Monte Carlo analysis of the effect of uncertainties on model-based HVAC fault detection and diagnostics. HVAC&R Res 2014;20:616–27.
- [203] Andriamamonjy A, Saelens D, Klein R. An auto-deployed model-based fault detection and diagnosis approach for air handling units using BIM and Modelica. Autom Constr 2018;96:508–26.
- [204] Lu Q, Xie X, Parlikad AK, Schooling JM. Digital twin-enabled anomaly detection for built asset monitoring in operation and maintenance. Autom Constr 2020;118:103277.
- [205] Sun B, Luh PB, Jia Q-S, O'Neill Z, Song F. Building energy doctors: An SPC and Kalman filter-based method for system-level fault detection in HVAC systems. IEEE Trans Autom Sci Eng 2013;11:215–29.
- [206] Guo F, Rasmussen BP. Fault detection and diagnosis for residential HVAC systems using transient cloud-based thermostat data. In: 6th international high performance buildings conference at purdue. 2021, p. 378–88.
- [207] Lin G, Kramer H, Granderson J. Building fault detection and diagnostics: Achieved savings, and methods to evaluate algorithm performance. Build Environ 2020;168:106505.
- [208] Deng X, Liu Q, Deng Y, Mahadevan S. An improved method to construct basic probability assignment based on the confusion matrix for classification problem. Inform Sci 2016;340:250–61.
- [209] Pillai I, Fumera G, Roli F. F-measure optimisation in multi-label classifiers. In: Proceedings of the 21st international conference on pattern recognition. IEEE; 2012, p. 2424–7.
- [210] Adler W, Lausen B. Bootstrap estimated true and false positive rates and ROC curve. Comput Statist Data Anal 2009;53:718–29.
- [211] McClish DK. Analyzing a portion of the ROC curve. Med Decis Making 1989;9:190–5.
- [212] Bradley AP. The use of the area under the ROC curve in the evaluation of machine learning algorithms. Pattern Recognit 1997;30:1145–59.
- [213] Kim TK. T test as a parametric statistic. Korean J Anesthesiol 2015;68:540.
- [214] Lachenbruch P. McNemar test. In: Wiley StatsRef: Statistics reference online.
- 2014.
 [215] Woolson RF. Wilcoxon signed-rank test. In: Wiley Encyclopedia of clinical trials. 2007, p. 1–3.
- [216] Friedman M. The use of ranks to avoid the assumption of normality implicit in the analysis of variance. J Amer Statist Assoc 1937;32:675–701.
- [217] Nemenyi PB. Distribution-free multiple comparisons. Princeton University; 1963.
- [218] Blanes L, Costa A, Réhault N, Keane MM. Integration of fault detection and diagnosis with energy management standard ISO 50001 and operations and maitenance of HVAC systems. In: CLIMA 2013. 2013.
- [219] Chen Y, Lin G, Chen Z, Wen J, Granderson J. A simulation-based evaluation of fan coil unit fault effects. Energy Build 2022;263:112041.
- [220] Frank S, Heaney M, Jin X, Robertson J, Cheung H, Elmore R, et al. Hybrid model-based and data-driven fault detection and diagnostics for commercial buildings. Technical Report, Golden, CO (United States): National Renewable Energy Lab. (NREL); 2016.
- [221] Pan SJ, Yang Q. A survey on transfer learning. IEEE Trans Knowl Data Eng 2009;22:1345–59.
- [222] Miyata S, Akashi Y, Kuwahara Y, Tanaka K. Improving the training efficiency of automated fault detection and diagnosis for central chilled water plants by transfer learning. In: Building simulation 2021. September 2021, Bruge, Belgium, 2021.
- [223] Fan C, Xiao F, Yan C, Liu C, Li Z, Wang J. A novel methodology to explain and evaluate data-driven building energy performance models based on interpretable machine learning. Appl Energy 2019;235:1551–60.
- [224] Peacock M. Anomaly detection in BACnet/IP managed building automation systems. 2019.
- [225] Fu Y, O'Neill Z, Yang Z, Adetola V, Wen J, Ren L, et al. Modeling and evaluation of cyber-attacks on grid-interactive efficient buildings. Appl Energy 2021;303:117639.
- [226] Fu Y, O'Neill Z, Wen J, Adetola V. Evaluating the impact of cyber-attacks on grid-interactive efficient buildings. In: ASME international mechanical engineering congress and exposition, vol. 85642. American Society of Mechanical Engineers; 2021, V08BT08A047.
- [227] Habibzadeh H, Nussbaum BH, Anjomshoa F, Kantarci B, Soyata T. A survey on cybersecurity, data privacy, and policy issues in cyber-physical system deployments in smart cities. Sustainable Cities Soc 2019;50:101660.
- [228] Gayeski N, Kleindienst S, Gagne J, Werntz B, Cruz R, Samouhos S. Data and interfaces for advanced building operations and maintenance. RP 1633 final report, Atlanta: ASHRAE; 2015.
- [229] Peng C, van Doorn J, Eggers F, Wieringa JE. The effect of required warmth on consumer acceptance of artificial intelligence in service: The moderating role of AI-human collaboration. Int J Inf Manage 2022;66:102533.